



DEPARTMENT OF HEALTH AND HUMAN SERVICES

NEW HAMPSHIRE HOSPITAL

Lori A. Shibinette Commissioner

Heather M. Moquin Chief Executive Officer 36 CLINTON STREET, CONCORD, NH 03301 603-271-5300 1-800-852-3345 Ext. 5300 Fax: 603-271-5395 TDD Access: 1-800-735-2964 www.dhhs.nh.gov

STATE OF NEW HAMPSHIRE

January 18, 2021

His Excellency, Governor Christopher T. Sununu and the Honorable Council State House Concord, New Hampshire 03301

REQUESTED ACTION

Authorize the Department of Health and Human Services, New Hampshire Hospital, to enter into a **Sole Source** contract with Bretton Healthcare Solutions, LLC (VC#389157), Concord, NH, in the amount of \$330,670 for recruiting permanent nursing staff for New Hampshire Hospital and Glencliff Home, with the option to renew for up to two (2) additional years, effective upon Governor and Council approval through June 30, 2023. 36.37% General Funds. 63.63% Other Funds (Provider Fees).

Funds are available in the following accounts for State Fiscal Year 2022 and 2023, with the authority to adjust budget line items within the price limitation through the Budget Office, if needed and justified.

05-95-94-940010-87500000 HHS: New Hampshire Hospital, New Hampshire Hospital, Acute Psychiatric Services

State Fiscal Year	Class / Account	Class Title	Job Number	Total Amount
2022	102-500731	Contracts for Prog Svc	94051700	\$71,885
2023	102-500731	Contracts for Prog Svc	94051700	\$143,770
			Subtotal	\$215,655

05-95-91-910010-57100000 HHS: Glencliff Home, Glencliff Home, Glencliff Home Professional Care

State Fiscal Year	Class / Account	Class Title	Job Number	Total Amount
2022	046-500462	Consultants	91000000	\$43,131
2023 046-50	046-500462	Consultants	91000000	\$71,884
			Subtotal	\$115,015
			Total	\$330,670

His Excellency, Governor Christopher T. Sununu and the Honorable Council Page 2 of 2

EXPLANATION

This request is **Sole Source** because there are no known viable alternatives to the services provided by the Contractor. No other vendor known to New Hampshire Hospital or Glencliff Home has been interested in providing permanent staffing recruitment services at this scale. Additionally, the vendor is local to the Concord area, further increasing their understanding of the local labor market as it pertains to New Hampshire.

The purpose of this request is to recruit permanent nursing staff for New Hampshire Hospital and Glencliff Home. Currently the Department has many contracts for travel nurses or other forms of contracted staff. However, a more cost-effective solution to position vacancies is to fill positions permanently. New Hampshire Hospital and Glencliff Home have been searching for a vendor to partner with in this regard for over a year.

The Department will only incur costs 60 days after the start date of a Registered Nurse or a Licensed Practical Nurse at New Hampshire Hospital or Glencliff Home. If recruited staff terminate prior to completing 90 days of employment with the Department, the Department can recoup payments for recruitment services for up to 66% of the fees paid based on the length of time the employee remained employed with the Department.

The Department will monitor services by:

- Monitoring the number of nurse candidates put forth to the Department for hiring consideration.
- Monitoring the number of nurse candidates successfully hired by the Department.
- Monitoring the number of successfully hired nurses who resign their positions for any reason or are terminated by the Department within the first six (6) months of employment.

As referenced in Exhibit A, Revisions to Standard Agreement Provisions of the attached agreement, the parties have the option to extend the agreement for up two (2) additional years, contingent upon satisfactory delivery of services, available funding, agreement of the parties and Governor and Council approval.

Should the Governor and Council not authorize this request New Hampshire Hospital and Glencliff Home will continue to rely on expensive, temporary contracted staff, which is not sustainable, and can be highly disruptive as contracted staff typically work in 13-week increments.

Area served: New Hampshire Hospital and Glencliff Home

In the event that the Other Funds become no longer available, General Funds will not be requested to support this program.

Respectfully submitted,

Joseph Caristi

Joseph T. Caristi —3FA927B12BA948F...

Chief Financial Officer

Subject:_Permanent Staffing Recruitment for Registered Nurses (SS-2022-NHH-12-PERMA-01)

Notice: This agreement and all of its attachments shall become public upon submission to Governor and Executive Council for approval. Any information that is private, confidential or proprietary must be clearly identified to the agency and agreed to in writing prior to signing the contract.

AGREEMENT

The State of New Hampshire and the Contractor hereby mutually agree as follows:

GENERAL PROVISIONS

1. IDENTIFICATION.

1.1 State Agency Name		1.2 State Agency Address	
New Hampshire Department of Health and Human Services		129 Pleasant Street Concord, NH 03301-3857	
1.3 Contractor Name		1.4 Contractor Address	
Bretton Healthcare Solutions, LLC		32 Foxcross Circle, Concor	rd, NH 03301
1.5 Contractor Phone Number	1.6 Account Number 094-8750-102-500731	1.7 Completion Date June 30, 2023	1.8 Price Limitation \$330,670
(603) 724-3901	091-5710-046-500462		4550,070
1.9 Contracting Officer for State	te Agency	1.10 State Agency Telephone N	umber
Nathan D. White, Director		(603) 271-9631	
1.11 Contractor Signature DocuSigned by:	3/22/2022	1.12 Name and Title of Contractor Signatory Greg Carbone	
Grey Carbone Date:		Principal	
1.13 State Agency Signature Docusigned by:			gency Signatory
Ellen Marie Lapointe Date:		Chief Executive Officer	- NHH
1.15 Approval by the N.H. Department of Administration, Division		ion of Personnel (if applicable)	
By:		Director, On:	
1.16 Approval by the Attorney General (Form, Substance and Execution) (if applicable)			
By: Polyn Quenno		On: 3/23/2022	
1.17 Approval by the Governor and Executive Council (if applicable)			
G&C Item number:		G&C Meeting Date:	

2. SERVICES TO BE PERFORMED. The State of New Hampshire, acting through the agency identified in block 1.1 ("State"), engages contractor identified in block 1.3 ("Contractor") to perform, and the Contractor shall perform, the work or sale of goods, or both, identified and more particularly described in the attached EXHIBIT B which is incorporated herein by reference ("Services").

3. EFFECTIVE DATE/COMPLETION OF SERVICES.

- 3.1 Notwithstanding any provision of this Agreement to the contrary, and subject to the approval of the Governor and Executive Council of the State of New Hampshire, if applicable, this Agreement, and all obligations of the parties hereunder, shall become effective on the date the Governor and Executive Council approve this Agreement as indicated in block 1.17, unless no such approval is required, in which case the Agreement shall become effective on the date the Agreement is signed by the State Agency as shown in block 1.13 ("Effective Date").
- 3.2 If the Contractor commences the Services prior to the Effective Date, all Services performed by the Contractor prior to the Effective Date shall be performed at the sole risk of the Contractor, and in the event that this Agreement does not become effective, the State shall have no liability to the Contractor, including without limitation, any obligation to pay the Contractor for any costs incurred or Services performed. Contractor must complete all Services by the Completion Date specified in block 1.7.

4. CONDITIONAL NATURE OF AGREEMENT.

Notwithstanding any provision of this Agreement to the contrary, all obligations of the State hereunder, including, without limitation, the continuance of payments hereunder, are contingent upon the availability and continued appropriation of funds affected by any state or federal legislative or executive action that reduces, eliminates or otherwise modifies the appropriation or availability of funding for this Agreement and the Scope for Services provided in EXHIBIT B, in whole or in part. In no event shall the State be liable for any payments hereunder in excess of such available appropriated funds. In the event of a reduction or termination of appropriated funds, the State shall have the right to withhold payment until such funds become available, if ever, and shall have the right to reduce or terminate the Services under this Agreement immediately upon giving the Contractor notice of such reduction or termination. The State shall not be required to transfer funds from any other account or source to the Account identified in block 1.6 in the event funds in that Account are reduced or unavailable.

5. CONTRACT PRICE/PRICE LIMITATION/PAYMENT.

- 5.1 The contract price, method of payment, and terms of payment are identified and more particularly described in EXHIBIT C which is incorporated herein by reference.
- 5.2 The payment by the State of the contract price shall be the only and the complete reimbursement to the Contractor for all expenses, of whatever nature incurred by the Contractor in the performance hereof, and shall be the only and the complete

compensation to the Contractor for the Services. The State shall have no liability to the Contractor other than the contract price.

- 5.3 The State reserves the right to offset from any amounts otherwise payable to the Contractor under this Agreement those liquidated amounts required or permitted by N.H. RSA 80:7 through RSA 80:7-c or any other provision of law.
- 5.4 Notwithstanding any provision in this Agreement to the contrary, and notwithstanding unexpected circumstances, in no event shall the total of all payments authorized, or actually made hereunder, exceed the Price Limitation set forth in block 1.8.

6. COMPLIANCE BY CONTRACTOR WITH LAWS AND REGULATIONS/ EQUAL EMPLOYMENT OPPORTUNITY.

- 6.1 In connection with the performance of the Services, the Contractor shall comply with all applicable statutes, laws, regulations, and orders of federal, state, county or municipal authorities which impose any obligation or duty upon the Contractor, including, but not limited to, civil rights and equal employment opportunity laws. In addition, if this Agreement is funded in any part by monies of the United States, the Contractor shall comply with all federal executive orders, rules, regulations and statutes, and with any rules, regulations and guidelines as the State or the United States issue to implement these regulations. The Contractor shall also comply with all applicable intellectual property laws.
- 6.2 During the term of this Agreement, the Contractor shall not discriminate against employees or applicants for employment because of race, color, religion, creed, age, sex, handicap, sexual orientation, or national origin and will take affirmative action to prevent such discrimination.
- 6.3. The Contractor agrees to permit the State or United States access to any of the Contractor's books, records and accounts for the purpose of ascertaining compliance with all rules, regulations and orders, and the covenants, terms and conditions of this Agreement.

7. PERSONNEL.

- 7.1 The Contractor shall at its own expense provide all personnel necessary to perform the Services. The Contractor warrants that all personnel engaged in the Services shall be qualified to perform the Services, and shall be properly licensed and otherwise authorized to do so under all applicable laws.
- 7.2 Unless otherwise authorized in writing, during the term of this Agreement, and for a period of six (6) months after the Completion Date in block 1.7, the Contractor shall not hire, and shall not permit any subcontractor or other person, firm or corporation with whom it is engaged in a combined effort to perform the Services to hire, any person who is a State employee or official, who is materially involved in the procurement, administration or performance of this Agreement. This provision shall survive termination of this Agreement.
- 7.3 The Contracting Officer specified in block 1.9, or his or her successor, shall be the State's representative. In the event of any dispute concerning the interpretation of this Agreement, the Contracting Officer's decision shall be final for the State.

8. EVENT OF DEFAULT/REMEDIES.

- 8.1 Any one or more of the following acts or omissions of the Contractor shall constitute an event of default hereunder ("Event of Default"):
- 8.1.1 failure to perform the Services satisfactorily or on schedule;
- 8.1.2 failure to submit any report required hereunder; and/or
- 8.1.3 failure to perform any other covenant, term or condition of this Agreement.
- 8.2 Upon the occurrence of any Event of Default, the State may take any one, or more, or all, of the following actions:
- 8.2.1 give the Contractor a written notice specifying the Event of Default and requiring it to be remedied within, in the absence of a greater or lesser specification of time, thirty (30) days from the date of the notice; and if the Event of Default is not timely cured, terminate this Agreement, effective two (2) days after giving the Contractor notice of termination:
- 8.2.2 give the Contractor a written notice specifying the Event of Default and suspending all payments to be made under this Agreement and ordering that the portion of the contract price which would otherwise accrue to the Contractor during the period from the date of such notice until such time as the State determines that the Contractor has cured the Event of Default shall never be paid to the Contractor;
- 8.2.3 give the Contractor a written notice specifying the Event of Default and set off against any other obligations the State may owe to the Contractor any damages the State suffers by reason of any Event of Default; and/or
- 8.2.4 give the Contractor a written notice specifying the Event of Default, treat the Agreement as breached, terminate the Agreement and pursue any of its remedies at law or in equity, or both.
- 8.3. No failure by the State to enforce any provisions hereof after any Event of Default shall be deemed a waiver of its rights with regard to that Event of Default, or any subsequent Event of Default. No express failure to enforce any Event of Default shall be deemed a waiver of the right of the State to enforce each and all of the provisions hereof upon any further or other Event of Default on the part of the Contractor.

9. TERMINATION.

- 9.1 Notwithstanding paragraph 8, the State may, at its sole discretion, terminate the Agreement for any reason, in whole or in part, by thirty (30) days written notice to the Contractor that the State is exercising its option to terminate the Agreement.
- 9.2 In the event of an early termination of this Agreement for any reason other than the completion of the Services, the Contractor shall, at the State's discretion, deliver to the Contracting Officer, not later than fifteen (15) days after the date of termination, a report ("Termination Report") describing in detail all Services performed, and the contract price earned, to and including the date of termination. The form, subject matter, content, and number of copies of the Termination Report shall be identical to those of any Final Report described in the attached EXHIBIT B. In addition, at the State's discretion, the Contractor shall, within 15 days of notice of early termination, develop and

submit to the State a Transition Plan for services under the Agreement.

10. DATA/ACCESS/CONFIDENTIALITY/PRESERVATION.

10.1 As used in this Agreement, the word "data" shall mean all information and things developed or obtained during the performance of, or acquired or developed by reason of, this Agreement, including, but not limited to, all studies, reports, files, formulae, surveys, maps, charts, sound recordings, video recordings, pictorial reproductions, drawings, analyses, graphic representations, computer programs, computer printouts, notes, letters, memoranda, papers, and documents, all whether finished or unfinished.

10.2 All data and any property which has been received from the State or purchased with funds provided for that purpose under this Agreement, shall be the property of the State, and shall be returned to the State upon demand or upon termination of this Agreement for any reason.

10.3 Confidentiality of data shall be governed by N.H. RSA chapter 91-A or other existing law. Disclosure of data requires prior written approval of the State.

11. CONTRACTOR'S RELATION TO THE STATE. In the performance of this Agreement the Contractor is in all respects an independent contractor, and is neither an agent nor an employee of the State. Neither the Contractor nor any of its officers, employees, agents or members shall have authority to bind the State or receive any benefits, workers' compensation or other emoluments provided by the State to its employees.

12. ASSIGNMENT/DELEGATION/SUBCONTRACTS.

- 12.1 The Contractor shall not assign, or otherwise transfer any interest in this Agreement without the prior written notice, which shall be provided to the State at least fifteen (15) days prior to the assignment, and a written consent of the State. For purposes of this paragraph, a Change of Control shall constitute assignment. "Change of Control" means (a) merger, consolidation, or a transaction or series of related transactions in which a third party, together with its affiliates, becomes the direct or indirect owner of fifty percent (50%) or more of the voting shares or similar equity interests, or combined voting power of the Contractor, or (b) the sale of all or substantially all of the assets of the Contractor.
- 12.2 None of the Services shall be subcontracted by the Contractor without prior written notice and consent of the State. The State is entitled to copies of all subcontracts and assignment agreements and shall not be bound by any provisions contained in a subcontract or an assignment agreement to which it is not a party.
- 13. INDEMNIFICATION. Unless otherwise exempted by law, the Contractor shall indemnify and hold harmless the State, its officers and employees, from and against any and all claims, liabilities and costs for any personal injury or property damages, patent or copyright infringement, or other claims asserted against the State, its officers or employees, which arise out of (or which may be claimed to arise out of) the acts or omipsions of the

Contractor Initials

Date^{3/22/2022}

Contractor, or subcontractors, including but not limited to the negligence, reckless or intentional conduct. The State shall not be liable for any costs incurred by the Contractor arising under this paragraph 13. Notwithstanding the foregoing, nothing herein contained shall be deemed to constitute a waiver of the sovereign immunity of the State, which immunity is hereby reserved to the State. This covenant in paragraph 13 shall survive the termination of this Agreement.

14. INSURANCE.

- 14.1 The Contractor shall, at its sole expense, obtain and continuously maintain in force, and shall require any subcontractor or assignee to obtain and maintain in force, the following insurance:
- 14.1.1 commercial general liability insurance against all claims of bodily injury, death or property damage, in amounts of not less than \$1,000,000 per occurrence and \$2,000,000 aggregate or excess; and
- 14.1.2 special cause of loss coverage form covering all property subject to subparagraph 10.2 herein, in an amount not less than 80% of the whole replacement value of the property.
- 14.2 The policies described in subparagraph 14.1 herein shall be on policy forms and endorsements approved for use in the State of New Hampshire by the N.H. Department of Insurance, and issued by insurers licensed in the State of New Hampshire.
- 14.3 The Contractor shall furnish to the Contracting Officer identified in block 1.9, or his or her successor, a certificate(s) of insurance for all insurance required under this Agreement. Contractor shall also furnish to the Contracting Officer identified in block 1.9, or his or her successor, certificate(s) of insurance for all renewal(s) of insurance required under this Agreement no later than ten (10) days prior to the expiration date of each insurance policy. The certificate(s) of insurance and any renewals thereof shall be attached and are incorporated herein by reference.

15. WORKERS' COMPENSATION.

- 15.1 By signing this agreement, the Contractor agrees, certifies and warrants that the Contractor is in compliance with or exempt from, the requirements of N.H. RSA chapter 281-A ("Workers' Compensation").
- 15.2 To the extent the Contractor is subject to the requirements of N.H. RSA chapter 281-A, Contractor shall maintain, and require any subcontractor or assignee to secure and maintain, payment of Workers' Compensation in connection with activities which the person proposes to undertake pursuant to this Agreement. The Contractor shall furnish the Contracting Officer identified in block 1.9, or his or her successor, proof of Workers' Compensation in the manner described in N.H. RSA chapter 281-A and any applicable renewal(s) thereof, which shall be attached and are incorporated herein by reference. The State shall not be responsible for payment of any Workers' Compensation premiums or for any other claim or benefit for Contractor, or any subcontractor or employee of Contractor, which might arise under applicable State of New Hampshire Workers' Compensation laws in connection with the performance of the Services under this Agreement.

- 16. NOTICE. Any notice by a party hereto to the other party shall be deemed to have been duly delivered or given at the time of mailing by certified mail, postage prepaid, in a United States Post Office addressed to the parties at the addresses given in blocks 1.2 and 1.4, herein.
- 17. AMENDMENT. This Agreement may be amended, waived or discharged only by an instrument in writing signed by the parties hereto and only after approval of such amendment, waiver or discharge by the Governor and Executive Council of the State of New Hampshire unless no such approval is required under the circumstances pursuant to State law, rule or policy.
- 18. CHOICE OF LAW AND FORUM. This Agreement shall be governed, interpreted and construed in accordance with the laws of the State of New Hampshire, and is binding upon and inures to the benefit of the parties and their respective successors and assigns. The wording used in this Agreement is the wording chosen by the parties to express their mutual intent, and no rule of construction shall be applied against or in favor of any party. Any actions arising out of this Agreement shall be brought and maintained in New Hampshire Superior Court which shall have exclusive jurisdiction thereof.
- 19. CONFLICTING TERMS. In the event of a conflict between the terms of this P-37 form (as modified in EXHIBIT A) and/or attachments and amendment thereof, the terms of the P-37 (as modified in EXHIBIT A) shall control.
- **20. THIRD PARTIES.** The parties hereto do not intend to benefit any third parties and this Agreement shall not be construed to confer any such benefit.
- **21. HEADINGS.** The headings throughout the Agreement are for reference purposes only, and the words contained therein shall in no way be held to explain, modify, amplify or aid in the interpretation, construction or meaning of the provisions of this Agreement.
- **22. SPECIAL PROVISIONS.** Additional or modifying provisions set forth in the attached EXHIBIT A are incorporated herein by reference.
- **23. SEVERABILITY.** In the event any of the provisions of this Agreement are held by a court of competent jurisdiction to be contrary to any state or federal law, the remaining provisions of this Agreement will remain in full force and effect.
- 24. ENTIRE AGREEMENT. This Agreement, which may be executed in a number of counterparts, each of which shall be deemed an original, constitutes the entire agreement and understanding between the parties, and supersedes all prior agreements and understandings with respect to the subject matter hereof.



Revisions to Standard Agreement Provisions

- 1. Revisions to Form P-37, General Provisions
 - 1.1. Paragraph 3, Effective Date/Completion of Services, is amended by adding subparagraph 3.3 as follows:
 - 3.3. The parties may extend the Agreement for up to two (2) additional year(s) from the Completion Date, contingent upon satisfactory delivery of services, available funding, agreement of the parties, and approval of the Governor and Executive Council.
 - 1.2. Paragraph 12, Assignment/Delegation/Subcontracts, is amended by adding subparagraph 12.3 as follows:
 - 12.3. Subcontractors are subject to the same contractual conditions as the Contractor and the Contractor is responsible to ensure subcontractor compliance with those conditions. The Contractor shall have written agreements with all subcontractors, specifying the work to be performed and how corrective action shall be managed if the subcontractor's performance is inadequate. The Contractor shall manage the subcontractor's performance on an ongoing basis and take corrective action as necessary. The Contractor shall annually provide the State with a list of all subcontractors provided for under this Agreement and notify the State of any inadequate subcontractor performance.

Contractor Initials

Date

Ds

3/11/2022

Scope of Services

1. Statement of Work

- 1.1. The Contractor shall provide permanent staffing recruitment services, utilizing best-practice strategies to secure employment of Licensed Practical Nurses (LPN) and Registered Nurses (RN) to New Hampshire Hospital (NHH) and Glencliff Home (GH). The Contractor shall ensure recruitment services include, but are not limited to:
 - 1.1.1. Implementing recruitment strategies, as approved by the Department.
 - 1.1.2. Creating and posting job advertisements in formats approved by the Department.
 - 1.1.3. Searching for potential candidates using hiring databases and social media, as approved by the Department.
 - 1.1.4. Conducting pre-employment screening of all candidates, which includes, but is not limited to review of:
 - 1.1.4.1. Licensure and/or certification.
 - 1.1.4.2. Resume.
 - 1.1.4.3. Experience.
 - 1.1.4.4. Appropriateness for:
 - 1.1.4.4.1. Representation by Contractor.
 - 1.1.4.4.2. Referral to the Department for employment consideration.
 - 1.1.5. Forwarding pre-employment screening documentation to the Department for verification prior to connecting any qualified candidate with the Department for consideration.
 - 1.1.6. Connecting qualified candidates with the Department for completion of a State job application and hiring consideration only after receiving confirmation from the Department that the pre-employment screening documentation is verified.
- 1.2. The Contractor shall ensure candidates are capable of duties that are within their scope of practice.
- 1.3. The Contractor shall begin recruiting RNs and LPNs, as identified by the Department, within 30 days of the contract effective date.

Contractor Initials

3/11/2022

Date

2. Exhibits Incorporated

2.1. The Contractor shall comply with all Exhibits C through J, which are attached hereto and incorporated by reference herein.

3. Reporting Requirements

- 3.1. The Contractor shall submit monthly reports to the Department within 15 days following the reporting period, to reflect services provided through this Agreement, which include, but are not limited to:
 - 3.1.1. Number of RN and LPN candidates recruited.
 - 3.1.2. Number of RN and LPN candidates put forth to the Department for hiring consideration.
 - 3.1.3. Number of RN and LPN candidates hired by the Department.
 - 3.1.4. Date of hire.
 - 3.1.5. Facility where the hired RN or LPN will be working.
 - 3.1.6. Number of successfully hired RNs and LPN's who resign his/her position for any reason or are terminated by the Department for reasonable cause within the first six (6) months of employment.

4. Performance Measures

- 4.1. The Contractor shall actively and regularly collaborate with the Department to enhance contract management, improve results, and adjust program delivery and policy based on successful outcomes.
- 4.2. The Contractor may be required to provide other key data and metrics to the Department, including client-level demographic, performance, and service data.
- 4.3. Where applicable, the Contractor shall collect and share data with the Department in a format specified by the Department.

5. Additional Terms

5.1. Impacts Resulting from Court Orders or Legislative Changes

5.1.1. The Contractor agrees that, to the extent future state or federal legislation or court orders may have an impact on the Services described herein, the State has the right to modify Service priorities and expenditure requirements under this Agreement so as to achieve compliance therewith.

5.2. Credits and Copyright Ownership

5.2.1. All documents, notices, press releases, research reports and other materials prepared during or resulting from the performance of the services of the Agreement shall include the following statement; The

Bretton Healthcare Solutions, LLC

Contractor Initials 3/11/2022
Date

preparation of this (report, document etc.) was financed under an Contract with the State of New Hampshire, Department of Health and Human Services, with funds provided in part by the State of New Hampshire and/or such other funding sources as were available or required, e.g., the United States Department of Health and Human Services."

- 5.2.2. All materials produced or purchased under the Agreement shall have prior approval from the Department before printing, production, distribution or use.
- 5.2.3. The Department shall retain copyright ownership for any and all original materials produced, including, but not limited to:
 - 5.2.3.1. Brochures.
 - 5.2.3.2. Resource directories.
 - 5.2.3.3. Protocols or guidelines.
 - 5.2.3.4. Posters.
 - 5.2.3.5. Reports.
- 5.2.4. The Contractor shall not reproduce any materials produced under the Agreement without prior written approval from the Department.

6. Records

- 6.1. The Contractor shall keep records that include, but are not limited to:
 - 6.1.1. Books, records, documents and other electronic or physical data evidencing and reflecting all costs and other expenses incurred by the Contractor in the performance of the Contract, and all income received or collected by the Contractor.
 - 6.1.2. All records must be maintained in accordance with accounting procedures and practices, which sufficiently and properly reflect all such costs and expenses, and which are acceptable to the Department, and to include, without limitation, all ledgers, books, records, and original evidence of costs such as purchase requisitions and orders, vouchers, requisitions for materials, inventories, valuations of in-kind contributions, labor time cards, payrolls, and other records requested or required by the Department.
- 6.2. During the term of this Agreement and the period for retention hereunder, the Department, the United States Department of Health and Human Services, and any of their designated representatives shall have access to all reports and records maintained pursuant to the Agreement for purposes of audit, examination, excerpts and transcripts. Upon the purchase by the Department of the maximum number of units provided for in the Agreement and purposes.

Contractor Initials 3/11/2022
Date

payment of the price limitation hereunder, the Agreement and all the obligations of the parties hereunder (except such obligations as, by the terms of the Agreement are to be performed after the end of the term of this Agreement and/or survive the termination of the Agreement) shall terminate, provided however, that if, upon review of the Final Expenditure Report the Department shall disallow any expenses claimed by the Contractor as costs hereunder the Department shall retain the right, at its discretion, to deduct the amount of such expenses as are disallowed or to recover such sums from the Contractor.

Contractor Initials

3/11/2022

Date

Payment Terms

- This Agreement is funded by:
 - 1.1. 36.37% General funds.
 - 1.2. 63.63% Other funds (Provider Fees).
- 2. For the purposes of this Agreement:
 - 2.1. The Department has identified the Contractor as a Contractor, in accordance with 2 CFR 200.331.
 - 2.2. The Department has identified this Agreement as NON-R&D, in accordance with 2 CFR §200.332.
- 3. Payment for services shall be in accordance with the rates identified below for positions successfully filled as indicated below:
 - 3.1. Payment for permanently placed, full-time, RNs shall be equal to 20% of the annualized earnings based on the hours and rate offered and accepted in their employment agreement.
 - 3.2. Payment for permanently placed, full-time, LPNs shall be equal to 20% of the annualized earnings based on the hours and rate offered and accepted in their employment agreement.
 - 3.3. Payment for part-time or per diem nurses shall be equal to 20% of the annualized earnings based on the hours and rate offered and accepted in their employment agreement.
- 4. The Contractor shall not invoice the Department for any referred RN or LPN who resigns or is terminated by the Department from the position for any reason within the first 30 days of employment.
- 5. The Contractor shall shall refund any fees paid by the Department for nurses who resign or are terminated by the Department in accordance with the terms for refunds as indicated below:
 - 5.1. If an employed RN or LPNresigns or is terminated by the the Department from their position, for any reason, between 31 and 60 days of employment, the Contractor shall refund the Department in an amount of 66% of any fee paid.
 - 5.2. If an employed RN or LPN resigns or is terminated by the the Department from their position, for any reason, between 61 and 90 days of employment, the Contractor shall refund the Department in an amount of 33% any fee paid.
- 6. The Contractor shall submit an invoice in a form satisfactory to the Department 45 days after the hire date of the candidate, which identifies and requests reimbursement for authorized expenses incurred in the prior month. Des The

Contractor Initials

3/22/2022

Date

GC

C-1.2

Contractor shall ensure the invoice is completed, dated and submitted to the Department in order to initiate payment.

7. In lieu of hard copies, all invoices may be assigned an electronic signature and emailed to NHHFinancialServices@dhhs.nh.gov, or invoices may be mailed to:

Financial Services – Accounts Payable NH Hospital - Department of Health and Human Services 121 South Fruit Street Concord, NH 03301

- 8. The Department shall make payment to the Contractor within thirty (30) days of receipt of each invoice, subsequent to approval of the submitted invoice and if sufficient funds are available, subject to Paragraph 4 of the General Provisions Form Number P-37 of this Agreement.
- 9. The final invoice shall be due to the Department no later than forty (40) days after the contract completion date specified in Form P-37, General Provisions Block 1.7 Completion Date.
- 10. The Contractor must provide the services in Exhibit B, Scope of Services, in compliance with funding requirements.
- 11. The Contractor agrees that funding under this Agreement may be withheld, in whole or in part in the event of non-compliance with the terms and conditions of Exhibit B, Scope of Services.
- 12. Notwithstanding anything to the contrary herein, the Contractor agrees that funding under this agreement may be withheld, in whole or in part, in the event of non-compliance with any Federal or State law, rule or regulation applicable to the services provided, or if the said services or products have not been satisfactorily completed in accordance with the terms and conditions of this agreement.
- 13. Notwithstanding Paragraph 17 of the General Provisions Form P-37, changes limited to adjusting amounts within the price limitation and adjusting encumbrances between State Fiscal Years and budget class lines through the Budget Office may be made by written agreement of both parties, without obtaining approval of the Governor and Executive Council, if needed and justified.
- 14. Audits
 - 14.1. The Contractor must email an annual audit to melissa.s.morin@dhhs.nh.gov if any of the following conditions exist:
 - 14.1.1. Condition A The Contractor expended \$750,000 or more in federal funds received as a subrecipient pursuant to 2 CFR Part 200, during the most recently completed fiscal year.

	DS
Contractor Initials	ac
Contractor initials	
	3/22/2022
Data	

Bretton Healthcare Solutions, LLC

SS-2022-NHH-12-PERMA-01

- 14.1.2. Condition B The Contractor is subject to audit pursuant to the requirements of NH RSA 7:28, III-b, pertaining to charitable organizations receiving support of \$1,000,000 or more.
- 14.1.3. Condition C The Contractor is a public company and required by Security and Exchange Commission (SEC) regulations to submit an annual financial audit.
- 14.2. If Condition A exists, the Contractor shall submit an annual single audit performed by an independent Certified Public Accountant (CPA) to the Department within 120 days after the close of the Contractor's fiscal year, conducted in accordance with the requirements of 2 CFR Part 200, Subpart F of the Uniform Administrative Requirements, Cost Principles, and Audit Requirements for Federal awards.
- 14.3. If Condition B or Condition C exists, the Contractor shall submit an annual financial audit performed by an independent CPA within 120 days after the close of the Contractor's fiscal year.
- 14.4. In addition to, and not in any way in limitation of obligations of the Contract, it is understood and agreed by the Contractor that the Contractor shall be held liable for any state or federal audit exceptions and shall return to the Department all payments made under the Contract to which exception has been taken, or which have been disallowed because of such an exception.

Contractor Initials

3/22/2022

Date



CERTIFICATION REGARDING DRUG-FREE WORKPLACE REQUIREMENTS

The Vendor identified in Section 1.3 of the General Provisions agrees to comply with the provisions of Sections 5151-5160 of the Drug-Free Workplace Act of 1988 (Pub. L. 100-690, Title V, Subtitle D; 41 U.S.C. 701 et seq.), and further agrees to have the Contractor's representative, as identified in Sections 1.11 and 1.12 of the General Provisions execute the following Certification:

ALTERNATIVE I - FOR GRANTEES OTHER THAN INDIVIDUALS

US DEPARTMENT OF HEALTH AND HUMAN SERVICES - CONTRACTORS
US DEPARTMENT OF EDUCATION - CONTRACTORS
US DEPARTMENT OF AGRICULTURE - CONTRACTORS

This certification is required by the regulations implementing Sections 5151-5160 of the Drug-Free Workplace Act of 1988 (Pub. L. 100-690, Title V, Subtitle D; 41 U.S.C. 701 et seq.). The January 31, 1989 regulations were amended and published as Part II of the May 25, 1990 Federal Register (pages 21681-21691), and require certification by grantees (and by inference, sub-grantees and sub-contractors), prior to award, that they will maintain a drug-free workplace. Section 3017.630(c) of the regulation provides that a grantee (and by inference, sub-grantees and sub-contractors) that is a State may elect to make one certification to the Department in each federal fiscal year in lieu of certificates for each grant during the federal fiscal year covered by the certification. The certificate set out below is a material representation of fact upon which reliance is placed when the agency awards the grant. False certification or violation of the certification shall be grounds for suspension of payments, suspension or termination of grants, or government wide suspension or debarment. Contractors using this form should send it to:

Commissioner NH Department of Health and Human Services 129 Pleasant Street, Concord, NH 03301-6505

- 1. The grantee certifies that it will or will continue to provide a drug-free workplace by:
 - 1.1. Publishing a statement notifying employees that the unlawful manufacture, distribution, dispensing, possession or use of a controlled substance is prohibited in the grantee's workplace and specifying the actions that will be taken against employees for violation of such prohibition:
 - 1.2. Establishing an ongoing drug-free awareness program to inform employees about
 - 1.2.1. The dangers of drug abuse in the workplace;
 - 1.2.2. The grantee's policy of maintaining a drug-free workplace;
 - 1.2.3. Any available drug counseling, rehabilitation, and employee assistance programs; and
 - 1.2.4. The penalties that may be imposed upon employees for drug abuse violations occurring in the workplace:
 - 1.3. Making it a requirement that each employee to be engaged in the performance of the grant be given a copy of the statement required by paragraph (a);
 - 1.4. Notifying the employee in the statement required by paragraph (a) that, as a condition of employment under the grant, the employee will
 - 1.4.1. Abide by the terms of the statement; and
 - 1.4.2. Notify the employer in writing of his or her conviction for a violation of a criminal drug statute occurring in the workplace no later than five calendar days after such conviction:
 - 1.5. Notifying the agency in writing, within ten calendar days after receiving notice under subparagraph 1.4.2 from an employee or otherwise receiving actual notice of such conviction. Employers of convicted employees must provide notice, including position title, to every grant officer on whose grant activity the convicted employee was working, unless the Federal agency

Vendor Initials

Date

3/11/2022



has designated a central point for the receipt of such notices. Notice shall include the identification number(s) of each affected grant;

- Taking one of the following actions, within 30 calendar days of receiving notice under subparagraph 1.4.2, with respect to any employee who is so convicted
 - 1.6.1. Taking appropriate personnel action against such an employee, up to and including termination, consistent with the requirements of the Rehabilitation Act of 1973, as amended; or
 - Requiring such employee to participate satisfactorily in a drug abuse assistance or rehabilitation program approved for such purposes by a Federal, State, or local health, law enforcement, or other appropriate agency;

Principal

- Making a good faith effort to continue to maintain a drug-free workplace through implementation of paragraphs 1.1, 1.2, 1.3, 1.4, 1.5, and 1.6.
- 2. The grantee may insert in the space provided below the site(s) for the performance of work done in connection with the specific grant.

Place of Performance (street address, city, county, state, zip code) (list each location)

Check □ if there are workplaces on file that are not identified here. Vendor Name: 3/11/2022 Name: Greg Carbone Date Title:

> Vendor Initials 3/11/2022 Date



CERTIFICATION REGARDING LOBBYING

The Vendor identified in Section 1.3 of the General Provisions agrees to comply with the provisions of Section 319 of Public Law 101-121, Government wide Guidance for New Restrictions on Lobbying, and 31 U.S.C. 1352, and further agrees to have the Contractor's representative, as identified in Sections 1.11 and 1.12 of the General Provisions execute the following Certification:

US DEPARTMENT OF HEALTH AND HUMAN SERVICES - CONTRACTORS
US DEPARTMENT OF EDUCATION - CONTRACTORS
US DEPARTMENT OF AGRICULTURE - CONTRACTORS

Programs (indicate applicable program covered):

- *Temporary Assistance to Needy Families under Title IV-A
- *Child Support Enforcement Program under Title IV-D
- *Social Services Block Grant Program under Title XX
- *Medicaid Program under Title XIX
- *Community Services Block Grant under Title VI
- *Child Care Development Block Grant under Title IV

The undersigned certifies, to the best of his or her knowledge and belief, that:

- 1. No Federal appropriated funds have been paid or will be paid by or on behalf of the undersigned, to any person for influencing or attempting to influence an officer or employee of any agency, a Member of Congress, an officer or employee of Congress, or an employee of a Member of Congress in connection with the awarding of any Federal contract, continuation, renewal, amendment, or modification of any Federal contract, grant, loan, or cooperative agreement (and by specific mention sub-grantee or sub-contractor).
- 2. If any funds other than Federal appropriated funds have been paid or will be paid to any person for influencing or attempting to influence an officer or employee of any agency, a Member of Congress, an officer or employee of Congress, or an employee of a Member of Congress in connection with this Federal contract, grant, loan, or cooperative agreement (and by specific mention sub-grantee or sub-contractor), the undersigned shall complete and submit Standard Form LLL, (Disclosure Form to Report Lobbying, in accordance with its instructions, attached and identified as Standard Exhibit E-I.)
- 3. The undersigned shall require that the language of this certification be included in the award document for sub-awards at all tiers (including subcontracts, sub-grants, and contracts under grants, loans, and cooperative agreements) and that all sub-recipients shall certify and disclose accordingly.

This certification is a material representation of fact upon which reliance was placed when this transaction was made or entered into. Submission of this certification is a prerequisite for making or entering into this transaction imposed by Section 1352, Title 31, U.S. Code. Any person who fails to file the required certification shall be subject to a civil penalty of not less than \$10,000 and not more than \$100,000 for each such failure.

Vendor Name:

3/11/2022	Orey Carbone	
Date	Name: Greg Carbone Title: Principal	
	Principal	
		GC
	Exhibit E – Certification Regarding Lobbying	Vendor Initials 2 (11 (202
CU/DHHS/110713	Page 1 of 1	Date3/11/202



CERTIFICATION REGARDING DEBARMENT, SUSPENSION AND OTHER RESPONSIBILITY MATTERS

The Contractor identified in Section 1.3 of the General Provisions agrees to comply with the provisions of Executive Office of the President, Executive Order 12549 and 45 CFR Part 76 regarding Debarment, Suspension, and Other Responsibility Matters, and further agrees to have the Contractor's representative, as identified in Sections 1.11 and 1.12 of the General Provisions execute the following Certification:

INSTRUCTIONS FOR CERTIFICATION

- 1. By signing and submitting this proposal (contract), the prospective primary participant is providing the certification set out below.
- 2. The inability of a person to provide the certification required below will not necessarily result in denial of participation in this covered transaction. If necessary, the prospective participant shall submit an explanation of why it cannot provide the certification. The certification or explanation will be considered in connection with the NH Department of Health and Human Services' (DHHS) determination whether to enter into this transaction. However, failure of the prospective primary participant to furnish a certification or an explanation shall disqualify such person from participation in this transaction.
- 3. The certification in this clause is a material representation of fact upon which reliance was placed when DHHS determined to enter into this transaction. If it is later determined that the prospective primary participant knowingly rendered an erroneous certification, in addition to other remedies available to the Federal Government, DHHS may terminate this transaction for cause or default.
- 4. The prospective primary participant shall provide immediate written notice to the DHHS agency to whom this proposal (contract) is submitted if at any time the prospective primary participant learns that its certification was erroneous when submitted or has become erroneous by reason of changed circumstances.
- 5. The terms "covered transaction," "debarred," "suspended," "ineligible," "lower tier covered transaction," "participant," "person," "primary covered transaction," "principal," "proposal," and "voluntarily excluded," as used in this clause, have the meanings set out in the Definitions and Coverage sections of the rules implementing Executive Order 12549: 45 CFR Part 76. See the attached definitions.
- 6. The prospective primary participant agrees by submitting this proposal (contract) that, should the proposed covered transaction be entered into, it shall not knowingly enter into any lower tier covered transaction with a person who is debarred, suspended, declared ineligible, or voluntarily excluded from participation in this covered transaction, unless authorized by DHHS.
- 7. The prospective primary participant further agrees by submitting this proposal that it will include the clause titled "Certification Regarding Debarment, Suspension, Ineligibility and Voluntary Exclusion Lower Tier Covered Transactions," provided by DHHS, without modification, in all lower tier covered transactions and in all solicitations for lower tier covered transactions.
- 8. A participant in a covered transaction may rely upon a certification of a prospective participant in a lower tier covered transaction that it is not debarred, suspended, ineligible, or involuntarily excluded from the covered transaction, unless it knows that the certification is erroneous. A participant may decide the method and frequency by which it determines the eligibility of its principals. Each participant may, but is not required to, check the Nonprocurement List (of excluded parties).
- 9. Nothing contained in the foregoing shall be construed to require establishment of a system of records in order to render in good faith the certification required by this clause. The knowledge and

Contractor Initials 3/11/2022



information of a participant is not required to exceed that which is normally possessed by a prudent person in the ordinary course of business dealings.

10. Except for transactions authorized under paragraph 6 of these instructions, if a participant in a covered transaction knowingly enters into a lower tier covered transaction with a person who is suspended, debarred, ineligible, or voluntarily excluded from participation in this transaction, in addition to other remedies available to the Federal government, DHHS may terminate this transaction for cause or default.

PRIMARY COVERED TRANSACTIONS

- 11. The prospective primary participant certifies to the best of its knowledge and belief, that it and its principals:
 - 11.1. are not presently debarred, suspended, proposed for debarment, declared ineligible, or voluntarily excluded from covered transactions by any Federal department or agency;
 - 11.2. have not within a three-year period preceding this proposal (contract) been convicted of or had a civil judgment rendered against them for commission of fraud or a criminal offense in connection with obtaining, attempting to obtain, or performing a public (Federal, State or local) transaction or a contract under a public transaction; violation of Federal or State antitrust statutes or commission of embezzlement, theft, forgery, bribery, falsification or destruction of records, making false statements, or receiving stolen property;
 - 11.3. are not presently indicted for otherwise criminally or civilly charged by a governmental entity (Federal, State or local) with commission of any of the offenses enumerated in paragraph (I)(b) of this certification; and
 - 11.4. have not within a three-year period preceding this application/proposal had one or more public transactions (Federal, State or local) terminated for cause or default.
- 12. Where the prospective primary participant is unable to certify to any of the statements in this certification, such prospective participant shall attach an explanation to this proposal (contract).

LOWER TIER COVERED TRANSACTIONS

- 13. By signing and submitting this lower tier proposal (contract), the prospective lower tier participant, as defined in 45 CFR Part 76, certifies to the best of its knowledge and belief that it and its principals:
 - 13.1. are not presently debarred, suspended, proposed for debarment, declared ineligible, or voluntarily excluded from participation in this transaction by any federal department or agency.
 - 13.2. where the prospective lower tier participant is unable to certify to any of the above, such prospective participant shall attach an explanation to this proposal (contract).
- 14. The prospective lower tier participant further agrees by submitting this proposal (contract) that it will include this clause entitled "Certification Regarding Debarment, Suspension, Ineligibility, and Voluntary Exclusion Lower Tier Covered Transactions," without modification in all lower tier covered transactions and in all solicitations for lower tier covered transactions.

Contractor Name:

	DocuSigned by:	
3/11/2022	Grey Carbone	
Date	Name:Greg Carbone	100.000.000.000.000.000
	Title: Principal	

Contractor Initials

Date

3/11/2022



CERTIFICATION OF COMPLIANCE WITH REQUIREMENTS PERTAINING TO FEDERAL NONDISCRIMINATION, EQUAL TREATMENT OF FAITH-BASED ORGANIZATIONS AND WHISTLEBLOWER PROTECTIONS

The Contractor identified in Section 1.3 of the General Provisions agrees by signature of the Contractor's representative as identified in Sections 1.11 and 1.12 of the General Provisions, to execute the following certification:

Contractor will comply, and will require any subgrantees or subcontractors to comply, with any applicable federal nondiscrimination requirements, which may include:

- the Omnibus Crime Control and Safe Streets Act of 1968 (42 U.S.C. Section 3789d) which prohibits recipients of federal funding under this statute from discriminating, either in employment practices or in the delivery of services or benefits, on the basis of race, color, religion, national origin, and sex. The Act requires certain recipients to produce an Equal Employment Opportunity Plan:
- the Juvenile Justice Delinquency Prevention Act of 2002 (42 U.S.C. Section 5672(b)) which adopts by reference, the civil rights obligations of the Safe Streets Act. Recipients of federal funding under this statute are prohibited from discriminating, either in employment practices or in the delivery of services or benefits, on the basis of race, color, religion, national origin, and sex. The Act includes Equal Employment Opportunity Plan requirements;
- the Civil Rights Act of 1964 (42 U.S.C. Section 2000d, which prohibits recipients of federal financial assistance from discriminating on the basis of race, color, or national origin in any program or activity);
- the Rehabilitation Act of 1973 (29 U.S.C. Section 794), which prohibits recipients of Federal financial assistance from discriminating on the basis of disability, in regard to employment and the delivery of services or benefits, in any program or activity;
- the Americans with Disabilities Act of 1990 (42 U.S.C. Sections 12131-34), which prohibits discrimination and ensures equal opportunity for persons with disabilities in employment, State and local government services, public accommodations, commercial facilities, and transportation;
- the Education Amendments of 1972 (20 U.S.C. Sections 1681, 1683, 1685-86), which prohibits discrimination on the basis of sex in federally assisted education programs;
- the Age Discrimination Act of 1975 (42 U.S.C. Sections 6106-07), which prohibits discrimination on the basis of age in programs or activities receiving Federal financial assistance. It does not include employment discrimination;
- 28 C.F.R. pt. 31 (U.S. Department of Justice Regulations OJJDP Grant Programs); 28 C.F.R. pt. 42 (U.S. Department of Justice Regulations - Nondiscrimination; Equal Employment Opportunity; Policies and Procedures); Executive Order No. 13279 (equal protection of the laws for faith-based and community organizations); Executive Order No. 13559, which provide fundamental principles and policy-making criteria for partnerships with faith-based and neighborhood organizations;
- 28 C.F.R. pt. 38 (U.S. Department of Justice Regulations Equal Treatment for Faith-Based Organizations); and Whistleblower protections 41 U.S.C. §4712 and The National Defense Authorization Act (NDAA) for Fiscal Year 2013 (Pub. L. 112-239, enacted January 2, 2013) the Pilot Program for Enhancement of Contract Employee Whistleblower Protections, which protects employees against reprisal for certain whistle blowing activities in connection with federal grants and contracts.

The certificate set out below is a material representation of fact upon which reliance is placed when the agency awards the grant. False certification or violation of the certification shall be grounds for suspension of payments, suspension or termination of grants, or government wide suspension or debarment.

Exhibit G

Contractor Initials

Certification of Compliance with requirements pertaining to Federal Nondiscrimination, Equal Treatment of Faith-Based Organizations

6/27/14 Rev 10/21/14

Page 1 of 2

3/11/2022 Date



In the event a Federal or State court or Federal or State administrative agency makes a finding of discrimination after a due process hearing on the grounds of race, color, religion, national origin, or sex against a recipient of funds, the recipient will forward a copy of the finding to the Office for Civil Rights, to the applicable contracting agency or division within the Department of Health and Human Services, and to the Department of Health and Human Services Office of the Ombudsman.

The Contractor identified in Section 1.3 of the General Provisions agrees by signature of the Contractor's representative as identified in Sections 1.11 and 1.12 of the General Provisions, to execute the following certification:

1. By signing and submitting this proposal (contract) the Contractor agrees to comply with the provisions indicated above.

Contractor Name: 3/11/2022 Name: Greg Carbone Date Title: Principal

Exhibit G

Contractor Initials

Certification of Compliance with requirements pertaining to Federal Nondiscrimination, Equal Treatment of Faith-Based Organizations and Whistleblower protections

6/27/14 Rev. 10/21/14

Page 2 of 2

3/11/2022 Date



CERTIFICATION REGARDING ENVIRONMENTAL TOBACCO SMOKE

Public Law 103-227, Part C - Environmental Tobacco Smoke, also known as the Pro-Children Act of 1994 (Act), requires that smoking not be permitted in any portion of any indoor facility owned or leased or contracted for by an entity and used routinely or regularly for the provision of health, day care, education, or library services to children under the age of 18, if the services are funded by Federal programs either directly or through State or local governments, by Federal grant, contract, loan, or loan guarantee. The law does not apply to children's services provided in private residences, facilities funded solely by Medicare or Medicaid funds, and portions of facilities used for inpatient drug or alcohol treatment. Failure to comply with the provisions of the law may result in the imposition of a civil monetary penalty of up to \$1000 per day and/or the imposition of an administrative compliance order on the responsible entity.

The Contractor identified in Section 1.3 of the General Provisions agrees, by signature of the Contractor's representative as identified in Section 1.11 and 1.12 of the General Provisions, to execute the following certification:

1. By signing and submitting this contract, the Contractor agrees to make reasonable efforts to comply with all applicable provisions of Public Law 103-227, Part C, known as the Pro-Children Act of 1994.

Contractor Name:

DocuSigned by:

Name: Greg Carbone

Title: Principal

Exhibit H – Certification Regarding Environmental Tobacco Smoke Page 1 of 1 Contractor Initials

Date

3/11/2022

3/11/2022

Date



Exhibit I

HEALTH INSURANCE PORTABILITY AND ACCOUNTABILITY ACT BUSINESS ASSOCIATE AGREEMENT

The Contractor identified in Section 1.3 of the General Provisions of the Agreement agrees to comply with the Health Insurance Portability and Accountability Act, Public Law 104-191 and with the Standards for Privacy and Security of Individually Identifiable Health Information, 45 CFR Parts 160 and 164 applicable to business associates. As defined herein, "Business Associate" shall mean the Contractor and subcontractors and agents of the Contractor that receive, use or have access to protected health information under this Agreement and "Covered Entity" shall mean the State of New Hampshire, Department of Health and Human Services.

(1) Definitions.

- a. "Breach" shall have the same meaning as the term "Breach" in section 164.402 of Title 45, Code of Federal Regulations.
- b. "Business Associate" has the meaning given such term in section 160.103 of Title 45, Code of Federal Regulations.
- <u>"Covered Entity"</u> has the meaning given such term in section 160.103 of Title 45, Code of Federal Regulations.
- d. "<u>Designated Record Set</u>" shall have the same meaning as the term "designated record set" in 45 CFR Section 164.501.
- e. "<u>Data Aggregation</u>" shall have the same meaning as the term "data aggregation" in 45 CFR Section 164.501.
- f. "<u>Health Care Operations</u>" shall have the same meaning as the term "health care operations" in 45 CFR Section 164.501.
- g. "HITECH Act" means the Health Information Technology for Economic and Clinical Health Act, TitleXIII, Subtitle D, Part 1 & 2 of the American Recovery and Reinvestment Act of 2009.
- h. "<u>HIPAA</u>" means the Health Insurance Portability and Accountability Act of 1996, Public Law 104-191 and the Standards for Privacy and Security of Individually Identifiable Health Information, 45 CFR Parts 160, 162 and 164 and amendments thereto.
- "Individual" shall have the same meaning as the term "individual" in 45 CFR Section 160.103 and shall include a person who qualifies as a personal representative in accordance with 45 CFR Section 164.501(g).
- j. "Privacy Rule" shall mean the Standards for Privacy of Individually Identifiable Health Information at 45 CFR Parts 160 and 164, promulgated under HIPAA by the United States Department of Health and Human Services.
- k. "Protected Health Information" shall have the same meaning as the term "protected health information" in 45 CFR Section 160.103, limited to the information created or received by Business Associate from or on behalf of Covered Entity.

3/2014

Exhibit I
Health Insurance Portability Act
Business Associate Agreement
Page 1 of 6



Exhibit I

- "Required by Law" shall have the same meaning as the term "required by law" in 45 CFR Section 164.103.
- m. "Secretary" shall mean the Secretary of the Department of Health and Human Services or his/her designee.
- n. "Security Rule" shall mean the Security Standards for the Protection of Electronic Protected Health Information at 45 CFR Part 164, Subpart C, and amendments thereto.
- o. "Unsecured Protected Health Information" means protected health information that is not secured by a technology standard that renders protected health information unusable, unreadable, or indecipherable to unauthorized individuals and is developed or endorsed by a standards developing organization that is accredited by the American National Standards Institute.
- p. Other Definitions All terms not otherwise defined herein shall have the meaning established under 45 C.F.R. Parts 160, 162 and 164, as amended from time to time, and the HITECH Act.

(2) Business Associate Use and Disclosure of Protected Health Information.

- a. Business Associate shall not use, disclose, maintain or transmit Protected Health Information (PHI) except as reasonably necessary to provide the services outlined under Exhibit A of the Agreement. Further, Business Associate, including but not limited to all its directors, officers, employees and agents, shall not use, disclose, maintain or transmit PHI in any manner that would constitute a violation of the Privacy and Security Rule.
- b. Business Associate may use or disclose PHI:
 - I. For the proper management and administration of the Business Associate;
 - II. As required by law, pursuant to the terms set forth in paragraph d. below; or
 - III. For data aggregation purposes for the health care operations of Covered Entity.
- c. To the extent Business Associate is permitted under the Agreement to disclose PHI to a third party, Business Associate must obtain, prior to making any such disclosure, (i) reasonable assurances from the third party that such PHI will be held confidentially and used or further disclosed only as required by law or for the purpose for which it was disclosed to the third party; and (ii) an agreement from such third party to notify Business Associate, in accordance with the HIPAA Privacy, Security, and Breach Notification Rules of any breaches of the confidentiality of the PHI, to the extent it has obtained knowledge of such breach.
- d. The Business Associate shall not, unless such disclosure is reasonably necessary to provide services under Exhibit A of the Agreement, disclose any PHI in response to a request for disclosure on the basis that it is required by law, without first notifying Covered Entity so that Covered Entity has an opportunity to object to the disclosure and to seek appropriate relief. If Covered Entity objects to such disclosure, the Business



Exhibit I

Associate shall refrain from disclosing the PHI until Covered Entity has exhausted all remedies.

e. If the Covered Entity notifies the Business Associate that Covered Entity has agreed to be bound by additional restrictions over and above those uses or disclosures or security safeguards of PHI pursuant to the Privacy and Security Rule, the Business Associate shall be bound by such additional restrictions and shall not disclose PHI in violation of such additional restrictions and shall abide by any additional security safeguards.

(3) Obligations and Activities of Business Associate.

- a. The Business Associate shall notify the Covered Entity's Privacy Officer immediately after the Business Associate becomes aware of any use or disclosure of protected health information not provided for by the Agreement including breaches of unsecured protected health information and/or any security incident that may have an impact on the protected health information of the Covered Entity.
- b. The Business Associate shall immediately perform a risk assessment when it becomes aware of any of the above situations. The risk assessment shall include, but not be limited to:
 - o The nature and extent of the protected health information involved, including the types of identifiers and the likelihood of re-identification;
 - The unauthorized person used the protected health information or to whom the disclosure was made;
 - o Whether the protected health information was actually acquired or viewed
 - The extent to which the risk to the protected health information has been mitigated.

The Business Associate shall complete the risk assessment within 48 hours of the breach and immediately report the findings of the risk assessment in writing to the Covered Entity.

- The Business Associate shall comply with all sections of the Privacy, Security, and Breach Notification Rule.
- d. Business Associate shall make available all of its internal policies and procedures, books and records relating to the use and disclosure of PHI received from, or created or received by the Business Associate on behalf of Covered Entity to the Secretary for purposes of determining Covered Entity's compliance with HIPAA and the Privacy and Security Rule.
- e. Business Associate shall require all of its business associates that receive, use or have access to PHI under the Agreement, to agree in writing to adhere to the same restrictions and conditions on the use and disclosure of PHI contained herein, including the duty to return or destroy the PHI as provided under Section 3 (I). The Covered Entity shall be considered a direct third party beneficiary of the Contractor's business associate agreements with Contractor's intended business associates, who will be receiving PHI



Exhibit I

pursuant to this Agreement, with rights of enforcement and indemnification from such business associates who shall be governed by standard Paragraph #13 of the standard contract provisions (P-37) of this Agreement for the purpose of use and disclosure of protected health information.

- f. Within five (5) business days of receipt of a written request from Covered Entity, Business Associate shall make available during normal business hours at its offices all records, books, agreements, policies and procedures relating to the use and disclosure of PHI to the Covered Entity, for purposes of enabling Covered Entity to determine Business Associate's compliance with the terms of the Agreement.
- g. Within ten (10) business days of receiving a written request from Covered Entity, Business Associate shall provide access to PHI in a Designated Record Set to the Covered Entity, or as directed by Covered Entity, to an individual in order to meet the requirements under 45 CFR Section 164.524.
- h. Within ten (10) business days of receiving a written request from Covered Entity for an amendment of PHI or a record about an individual contained in a Designated Record Set, the Business Associate shall make such PHI available to Covered Entity for amendment and incorporate any such amendment to enable Covered Entity to fulfill its obligations under 45 CFR Section 164.526.
- Business Associate shall document such disclosures of PHI and information related to such disclosures as would be required for Covered Entity to respond to a request by an individual for an accounting of disclosures of PHI in accordance with 45 CFR Section 164.528.
- j. Within ten (10) business days of receiving a written request from Covered Entity for a request for an accounting of disclosures of PHI, Business Associate shall make available to Covered Entity such information as Covered Entity may require to fulfill its obligations to provide an accounting of disclosures with respect to PHI in accordance with 45 CFR Section 164.528.
- k. In the event any individual requests access to, amendment of, or accounting of PHI directly from the Business Associate, the Business Associate shall within two (2) business days forward such request to Covered Entity. Covered Entity shall have the responsibility of responding to forwarded requests. However, if forwarding the individual's request to Covered Entity would cause Covered Entity or the Business Associate to violate HIPAA and the Privacy and Security Rule, the Business Associate shall instead respond to the individual's request as required by such law and notify Covered Entity of such response as soon as practicable.
- I. Within ten (10) business days of termination of the Agreement, for any reason, the Business Associate shall return or destroy, as specified by Covered Entity, all PHI received from, or created or received by the Business Associate in connection with the Agreement, and shall not retain any copies or back-up tapes of such PHI. If return or destruction is not feasible, or the disposition of the PHI has been otherwise agreed to in the Agreement, Business Associate shall continue to extend the protections of the Agreement, to such PHI and limit further uses and disclosures of such PHI to the purposes that make the return or destruction infeasible, for so long as Business

Page 4 of 6



Exhibit I

Associate maintains such PHI. If Covered Entity, in its sole discretion, requires that the Business Associate destroy any or all PHI, the Business Associate shall certify to Covered Entity that the PHI has been destroyed.

(4) Obligations of Covered Entity

- a. Covered Entity shall notify Business Associate of any changes or limitation(s) in its Notice of Privacy Practices provided to individuals in accordance with 45 CFR Section 164.520, to the extent that such change or limitation may affect Business Associate's use or disclosure of PHI.
- b. Covered Entity shall promptly notify Business Associate of any changes in, or revocation of permission provided to Covered Entity by individuals whose PHI may be used or disclosed by Business Associate under this Agreement, pursuant to 45 CFR Section 164.506 or 45 CFR Section 164.508.
- c. Covered entity shall promptly notify Business Associate of any restrictions on the use or disclosure of PHI that Covered Entity has agreed to in accordance with 45 CFR 164.522, to the extent that such restriction may affect Business Associate's use or disclosure of PHI.

(5) Termination for Cause

In addition to Paragraph 10 of the standard terms and conditions (P-37) of this Agreement the Covered Entity may immediately terminate the Agreement upon Covered Entity's knowledge of a breach by Business Associate of the Business Associate Agreement set forth herein as Exhibit I. The Covered Entity may either immediately terminate the Agreement or provide an opportunity for Business Associate to cure the alleged breach within a timeframe specified by Covered Entity. If Covered Entity determines that neither termination nor cure is feasible, Covered Entity shall report the violation to the Secretary.

(6) Miscellaneous

- a. <u>Definitions and Regulatory References</u>. All terms used, but not otherwise defined herein, shall have the same meaning as those terms in the Privacy and Security Rule, amended from time to time. A reference in the Agreement, as amended to include this Exhibit I, to a Section in the Privacy and Security Rule means the Section as in effect or as amended.
- b. <u>Amendment</u>. Covered Entity and Business Associate agree to take such action as is necessary to amend the Agreement, from time to time as is necessary for Covered Entity to comply with the changes in the requirements of HIPAA, the Privacy and Security Rule, and applicable federal and state law.
- c. <u>Data Ownership</u>. The Business Associate acknowledges that it has no ownership rights with respect to the PHI provided by or created on behalf of Covered Entity.
- d. <u>Interpretation</u>. The parties agree that any ambiguity in the Agreement shall be resolved to permit Covered Entity to comply with HIPAA, the Privacy and Security Rule.

3/2014



Exhibit I

- e. <u>Segregation</u>. If any term or condition of this Exhibit I or the application thereof to any person(s) or circumstance is held invalid, such invalidity shall not affect other terms or conditions which can be given effect without the invalid term or condition; to this end the terms and conditions of this Exhibit I are declared severable.
- f. <u>Survival</u>. Provisions in this Exhibit I regarding the use and disclosure of PHI, return or destruction of PHI, extensions of the protections of the Agreement in section (3) I, the defense and indemnification provisions of section (3) e and Paragraph 13 of the standard terms and conditions (P-37), shall survive the termination of the Agreement.

IN WITNESS WHEREOF, the parties hereto have duly executed this Exhibit I.

Department of Health and Human Services	Bretton Healthcare Solutions LLC
The State by:	Names of the Contractor
Ellen Marie Lapointe	Grey Carbone
Signature of Authorized Representative	Signature of Authorized Representative
Ellen Marie Lapointe	Greg Carbone
Name of Authorized Representative Chief Operating Officer - NHH	Name of Authorized Representative
SSTEMBER (SERVE SHEET STEELER TOP COMPANY SHEET STEELER STEEL	Principal
Title of Authorized Representative	Title of Authorized Representative
3/14/2022	3/11/2022
Date	Date



CERTIFICATION REGARDING THE FEDERAL FUNDING ACCOUNTABILITY AND TRANSPARENCY ACT (FFATA) COMPLIANCE

The Federal Funding Accountability and Transparency Act (FFATA) requires prime awardees of individual Federal grants equal to or greater than \$25,000 and awarded on or after October 1, 2010, to report on data related to executive compensation and associated first-tier sub-grants of \$25,000 or more. If the initial award is below \$25,000 but subsequent grant modifications result in a total award equal to or over \$25,000, the award is subject to the FFATA reporting requirements, as of the date of the award. In accordance with 2 CFR Part 170 (Reporting Subaward and Executive Compensation Information), the Department of Health and Human Services (DHHS) must report the following information for any subaward or contract award subject to the FFATA reporting requirements:

- 1. Name of entity
- 2. Amount of award
- 3. Funding agency
- 4. NAICS code for contracts / CFDA program number for grants
- 5. Program source
- 6. Award title descriptive of the purpose of the funding action
- 7. Location of the entity
- 8. Principle place of performance
- 9. Unique identifier of the entity (DUNS #)
- 10. Total compensation and names of the top five executives if:
 - 10.1. More than 80% of annual gross revenues are from the Federal government, and those revenues are greater than \$25M annually and
 - 10.2. Compensation information is not already available through reporting to the SEC.

Prime grant recipients must submit FFATA required data by the end of the month, plus 30 days, in which the award or award amendment is made.

The Contractor identified in Section 1.3 of the General Provisions agrees to comply with the provisions of The Federal Funding Accountability and Transparency Act, Public Law 109-282 and Public Law 110-252, and 2 CFR Part 170 (Reporting Subaward and Executive Compensation Information), and further agrees to have the Contractor's representative, as identified in Sections 1.11 and 1.12 of the General Provisions execute the following Certification:

The below named Contractor agrees to provide needed information as outlined above to the NH Department of Health and Human Services and to comply with all applicable provisions of the Federal Financial Accountability and Transparency Act.

Contractor Name:

	DocuSigned by:	
3/11/2022	Grey Carbone	
Date	Name: Grego Carbone	
	Title: Principal	

Contractor Initials

Date

Date



FORM A

	ow listed questions are true and accurate.	General Provisions, I certify that the responses to the	
1.	The DUNS number for your entity is:		
2.	In your business or organization's preceding completed fiscal year, did your business or organization receive (1) 80 percent or more of your annual gross revenue in U.S. federal contracts, subcontract loans, grants, sub-grants, and/or cooperative agreements; and (2) \$25,000,000 or more in annual gross revenues from U.S. federal contracts, subcontracts, loans, grants, subgrants, and/or cooperative agreements?		
	If the answer to #2 above is NO, stop here		
	If the answer to #2 above is YES, please an	swer the following:	
3. Does the public have access to information about the compensation of the executives in your business or organization through periodic reports filed under section 13(a) or 15(d) of the Secur Exchange Act of 1934 (15 U.S.C.78m(a), 78o(d)) or section 6104 of the Internal Revenue Code 1986?			
	NOYES		
	If the answer to #3 above is YES, stop here		
	If the answer to #3 above is NO, please ans	swer the following:	
4.	The names and compensation of the five morganization are as follows:	ost highly compensated officers in your business or	
	Name:	Amount:	



DHHS Information Security Requirements

A. Definitions

The following terms may be reflected and have the described meaning in this document:

- 1. "Breach" means the loss of control, compromise, unauthorized disclosure, unauthorized acquisition, unauthorized access, or any similar term referring to situations where persons other than authorized users and for an other than authorized purpose have access or potential access to personally identifiable information, whether physical or electronic. With regard to Protected Health Information, "Breach" shall have the same meaning as the term "Breach" in section 164.402 of Title 45, Code of Federal Regulations.
- 2. "Computer Security Incident" shall have the same meaning "Computer Security Incident" in section two (2) of NIST Publication 800-61, Computer Security Incident Handling Guide, National Institute of Standards and Technology, U.S. Department of Commerce.
- 3. "Confidential Information" or "Confidential Data" means all confidential information disclosed by one party to the other such as all medical, health, financial, public assistance benefits and personal information including without limitation, Substance Abuse Treatment Records, Case Records, Protected Health Information and Personally Identifiable Information.

Confidential Information also includes any and all information owned or managed by the State of NH - created, received from or on behalf of the Department of Health and Human Services (DHHS) or accessed in the course of performing contracted services - of which collection, disclosure, protection, and disposition is governed by state or federal law or regulation. This information includes, but is not limited to Protected Health Information (PHI), Personal Information (PI), Personal Financial Information (PFI), Federal Tax Information (FTI), Social Security Numbers (SSN), Payment Card Industry (PCI), and or other sensitive and confidential information.

- 4. "End User" means any person or entity (e.g., contractor, contractor's employee, business associate, subcontractor, other downstream user, etc.) that receives DHHS data or derivative data in accordance with the terms of this Contract.
- 5. "HIPAA" means the Health Insurance Portability and Accountability Act of 1996 and the regulations promulgated thereunder.
- 6. "Incident" means an act that potentially violates an explicit or implied security policy, which includes attempts (either failed or successful) to gain unauthorized access to a system or its data, unwanted disruption or denial of service, the unauthorized use of a system for the processing or storage of data; and changes to system hardware, firmware, or software characteristics without the owner's knowledge, instruction, or consent. Incidents include the loss of data through theft or device misplacement, loss or misplacement of hardcopy documents, and misrouting of physical or electronic



DHHS Information Security Requirements

mail, all of which may have the potential to put the data at risk of unauthorized access, use, disclosure, modification or destruction.

- 7. "Open Wireless Network" means any network or segment of a network that is not designated by the State of New Hampshire's Department of Information Technology or delegate as a protected network (designed, tested, and approved, by means of the State, to transmit) will be considered an open network and not adequately secure for the transmission of unencrypted PI, PFI, PHI or confidential DHHS data.
- 8. "Personal Information" (or "PI") means information which can be used to distinguish or trace an individual's identity, such as their name, social security number, personal information as defined in New Hampshire RSA 359-C:19, biometric records, etc., alone, or when combined with other personal or identifying information which is linked or linkable to a specific individual, such as date and place of birth, mother's maiden name, etc.
- 9. "Privacy Rule" shall mean the Standards for Privacy of Individually Identifiable Health Information at 45 C.F.R. Parts 160 and 164, promulgated under HIPAA by the United States Department of Health and Human Services.
- 10. "Protected Health Information" (or "PHI") has the same meaning as provided in the definition of "Protected Health Information" in the HIPAA Privacy Rule at 45 C.F.R. § 160.103.
- 11. "Security Rule" shall mean the Security Standards for the Protection of Electronic Protected Health Information at 45 C.F.R. Part 164, Subpart C, and amendments thereto.
- 12. "Unsecured Protected Health Information" means Protected Health Information that is not secured by a technology standard that renders Protected Health Information unusable, unreadable, or indecipherable to unauthorized individuals and is developed or endorsed by a standards developing organization that is accredited by the American National Standards Institute.

I. RESPONSIBILITIES OF DHHS AND THE CONTRACTOR

- A. Business Use and Disclosure of Confidential Information.
 - The Contractor must not use, disclose, maintain or transmit Confidential Information except as reasonably necessary as outlined under this Contract. Further, Contractor, including but not limited to all its directors, officers, employees and agents, must not use, disclose, maintain or transmit PHI in any manner that would constitute a violation of the Privacy and Security Rule.
 - 2. The Contractor must not disclose any Confidential Information in response to a

Contractor Initials _____

V5. Last update 10/09/18

Exhibit K
DHHS Information
Security Requirements
Page 2 of 9

3/11/2022 Date ____



DHHS Information Security Requirements

request for disclosure on the basis that it is required by law, in response to a subpoena, etc., without first notifying DHHS so that DHHS has an opportunity to consent or object to the disclosure.

- 3. If DHHS notifies the Contractor that DHHS has agreed to be bound by additional restrictions over and above those uses or disclosures or security safeguards of PHI pursuant to the Privacy and Security Rule, the Contractor must be bound by such additional restrictions and must not disclose PHI in violation of such additional restrictions and must abide by any additional security safeguards.
- 4. The Contractor agrees that DHHS Data or derivative there from disclosed to an End User must only be used pursuant to the terms of this Contract.
- 5. The Contractor agrees DHHS Data obtained under this Contract may not be used for any other purposes that are not indicated in this Contract.
- The Contractor agrees to grant access to the data to the authorized representatives of DHHS for the purpose of inspecting to confirm compliance with the terms of this Contract.

II. METHODS OF SECURE TRANSMISSION OF DATA

- 1. Application Encryption. If End User is transmitting DHHS data containing Confidential Data between applications, the Contractor attests the applications have been evaluated by an expert knowledgeable in cyber security and that said application's encryption capabilities ensure secure transmission via the internet.
- Computer Disks and Portable Storage Devices. End User may not use computer disks or portable storage devices, such as a thumb drive, as a method of transmitting DHHS data.
- 3. Encrypted Email. End User may only employ email to transmit Confidential Data if email is <u>encrypted</u> and being sent to and being received by email addresses of persons authorized to receive such information.
- 4. Encrypted Web Site. If End User is employing the Web to transmit Confidential Data, the secure socket layers (SSL) must be used and the web site must be secure. SSL encrypts data transmitted via a Web site.
- 5. File Hosting Services, also known as File Sharing Sites. End User may not use file hosting services, such as Dropbox or Google Cloud Storage, to transmit Confidential Data.
- 6. Ground Mail Service. End User may only transmit Confidential Data via *certified* ground mail within the continental U.S. and when sent to a named individual.
- 7. Laptops and PDA. If End User is employing portable devices to transmit Confidential Data said devices must be encrypted and password-protected.
- 8. Open Wireless Networks. End User may not transmit Confidential Data via an open

Contractor Initials _____

Exhibit K
DHHS Information
Security Requirements
Page 3 of 9



DHHS Information Security Requirements

- wireless network. End User must employ a virtual private network (VPN) when remotely transmitting via an open wireless network.
- Remote User Communication. If End User is employing remote communication to access or transmit Confidential Data, a virtual private network (VPN) must be installed on the End User's mobile device(s) or laptop from which information will be transmitted or accessed.
- 10. SSH File Transfer Protocol (SFTP), also known as Secure File Transfer Protocol. If End User is employing an SFTP to transmit Confidential Data, End User will structure the Folder and access privileges to prevent inappropriate disclosure of information. SFTP folders and sub-folders used for transmitting Confidential Data will be coded for 24-hour auto-deletion cycle (i.e. Confidential Data will be deleted every 24 hours).
- 11. Wireless Devices. If End User is transmitting Confidential Data via wireless devices, all data must be encrypted to prevent inappropriate disclosure of information.

III. RETENTION AND DISPOSITION OF IDENTIFIABLE RECORDS

The Contractor will only retain the data and any derivative of the data for the duration of this Contract. After such time, the Contractor will have 30 days to destroy the data and any derivative in whatever form it may exist, unless, otherwise required by law or permitted under this Contract. To this end, the parties must:

A. Retention

- The Contractor agrees it will not store, transfer or process data collected in connection with the services rendered under this Contract outside of the United States. This physical location requirement shall also apply in the implementation of cloud computing, cloud service or cloud storage capabilities, and includes backup data and Disaster Recovery locations.
- 2. The Contractor agrees to ensure proper security monitoring capabilities are in place to detect potential security events that can impact State of NH systems and/or Department confidential information for contractor provided systems.
- 3. The Contractor agrees to provide security awareness and education for its End Users in support of protecting Department confidential information.
- 4. The Contractor agrees to retain all electronic and hard copies of Confidential Data in a secure location and identified in section IV. A.2
- 5. The Contractor agrees Confidential Data stored in a Cloud must be in a FedRAMP/HITECH compliant solution and comply with all applicable statutes and regulations regarding the privacy and security. All servers and devices must have currently-supported and hardened operating systems, the latest anti-viral, antihacker, anti-spam, anti-spyware, and anti-malware utilities. The environment, as a

Contractor Initials _____

Exhibit K DHHS Information Security Requirements

Page 4 of 9



DHHS Information Security Requirements

whole, must have aggressive intrusion-detection and firewall protection.

6. The Contractor agrees to and ensures its complete cooperation with the State's Chief Information Officer in the detection of any security vulnerability of the hosting infrastructure.

B. Disposition

- 1. If the Contractor will maintain any Confidential Information on its systems (or its sub-contractor systems), the Contractor will maintain a documented process for securely disposing of such data upon request or contract termination; and will obtain written certification for any State of New Hampshire data destroyed by the Contractor or any subcontractors as a part of ongoing, emergency, and or disaster recovery operations. When no longer in use, electronic media containing State of New Hampshire data shall be rendered unrecoverable via a secure wipe program in accordance with industry-accepted standards for secure deletion and media sanitization, or otherwise physically destroying the media (for example, degaussing) as described in NIST Special Publication 800-88, Rev 1, Guidelines for Media Sanitization, National Institute of Standards and Technology, U. S. Department of Commerce. The Contractor will document and certify in writing at time of the data destruction, and will provide written certification to the Department upon request. The written certification will include all details necessary to demonstrate data has been properly destroyed and validated. Where applicable, regulatory and professional standards for retention requirements will be jointly evaluated by the State and Contractor prior to destruction.
- Unless otherwise specified, within thirty (30) days of the termination of this Contract, Contractor agrees to destroy all hard copies of Confidential Data using a secure method such as shredding.
- 3. Unless otherwise specified, within thirty (30) days of the termination of this Contract, Contractor agrees to completely destroy all electronic Confidential Data by means of data erasure, also known as secure data wiping.

IV. PROCEDURES FOR SECURITY

- A. Contractor agrees to safeguard the DHHS Data received under this Contract, and any derivative data or files, as follows:
 - The Contractor will maintain proper security controls to protect Department confidential information collected, processed, managed, and/or stored in the delivery of contracted services.
 - 2. The Contractor will maintain policies and procedures to protect Department confidential information throughout the information lifecycle, where applicable, (from creation, transformation, use, storage and secure destruction) regardless of the media used to store the data (i.e., tape, disk, paper, etc.).

Contractor Initials

V5. Last update 10/09/18

Exhibit K
DHHS Information
Security Requirements
Page 5 of 9

3/11/2022 Date ____



DHHS Information Security Requirements

- 3. The Contractor will maintain appropriate authentication and access controls to contractor systems that collect, transmit, or store Department confidential information where applicable.
- 4. The Contractor will ensure proper security monitoring capabilities are in place to detect potential security events that can impact State of NH systems and/or Department confidential information for contractor provided systems.
- 5. The Contractor will provide regular security awareness and education for its End Users in support of protecting Department confidential information.
- 6. If the Contractor will be sub-contracting any core functions of the engagement supporting the services for State of New Hampshire, the Contractor will maintain a program of an internal process or processes that defines specific security expectations, and monitoring compliance to security requirements that at a minimum match those for the Contractor, including breach notification requirements.
- 7. The Contractor will work with the Department to sign and comply with all applicable State of New Hampshire and Department system access and authorization policies and procedures, systems access forms, and computer use agreements as part of obtaining and maintaining access to any Department system(s). Agreements will be completed and signed by the Contractor and any applicable sub-contractors prior to system access being authorized.
- 8. If the Department determines the Contractor is a Business Associate pursuant to 45 CFR 160.103, the Contractor will execute a HIPAA Business Associate Agreement (BAA) with the Department and is responsible for maintaining compliance with the agreement.
- 9. The Contractor will work with the Department at its request to complete a System Management Survey. The purpose of the survey is to enable the Department and Contractor to monitor for any changes in risks, threats, and vulnerabilities that may occur over the life of the Contractor engagement. The survey will be completed annually, or an alternate time frame at the Departments discretion with agreement by the Contractor, or the Department may request the survey be completed when the scope of the engagement between the Department and the Contractor changes.
- 10. The Contractor will not store, knowingly or unknowingly, any State of New Hampshire or Department data offshore or outside the boundaries of the United States unless prior express written consent is obtained from the Information Security Office leadership member within the Department.
- 11. Data Security Breach Liability. In the event of any security breach Contractor shall make efforts to investigate the causes of the breach, promptly take measures to prevent future breach and minimize any damage or loss resulting from the breach. The State shall recover from the Contractor all costs of response and recovery from

Contractor Initials _____

V5. Last update 10/09/18

Exhibit K
DHHS Information
Security Requirements
Page 6 of 9

3/11/2022 Date ____

New Hampshire Department of Health and Human Services Exhibit K



DHHS Information Security Requirements

the breach, including but not limited to: credit monitoring services, mailing costs and costs associated with website and telephone call center services necessary due to the breach.

- 12. Contractor must, comply with all applicable statutes and regulations regarding the privacy and security of Confidential Information, and must in all other respects maintain the privacy and security of PI and PHI at a level and scope that is not less than the level and scope of requirements applicable to federal agencies, including, but not limited to, provisions of the Privacy Act of 1974 (5 U.S.C. § 552a), DHHS Privacy Act Regulations (45 C.F.R. §5b), HIPAA Privacy and Security Rules (45 C.F.R. Parts 160 and 164) that govern protections for individually identifiable health information and as applicable under State law.
- 13. Contractor agrees to establish and maintain appropriate administrative, technical, and physical safeguards to protect the confidentiality of the Confidential Data and to prevent unauthorized use or access to it. The safeguards must provide a level and scope of security that is not less than the level and scope of security requirements established by the State of New Hampshire, Department of Information Technology. Refer to Vendor Resources/Procurement at https://www.nh.gov/doit/vendor/index.htm for the Department of Information Technology policies, guidelines, standards, and procurement information relating to vendors.
- 14. Contractor agrees to maintain a documented breach notification and incident response process. The Contractor will notify the State's Privacy Officer and the State's Security Officer of any security breach immediately, at the email addresses provided in Section VI. This includes a confidential information breach, computer security incident, or suspected breach which affects or includes any State of New Hampshire systems that connect to the State of New Hampshire network.
- 15. Contractor must restrict access to the Confidential Data obtained under this Contract to only those authorized End Users who need such DHHS Data to perform their official duties in connection with purposes identified in this Contract.
- 16. The Contractor must ensure that all End Users:
 - a. comply with such safeguards as referenced in Section IV A. above, implemented to protect Confidential Information that is furnished by DHHS under this Contract from loss, theft or inadvertent disclosure.
 - b. safeguard this information at all times.
 - c. ensure that laptops and other electronic devices/media containing PHI, PI, or PFI are encrypted and password-protected.
 - d. send emails containing Confidential Information only if <u>encrypted</u> and being sent to and being received by email addresses of persons authorized to receive such information.

Contractor Initials GC

V5. Last update 10/09/18

Exhibit K
DHHS Information
Security Requirements
Page 7 of 9

3/11/2022 Date

New Hampshire Department of Health and Human Services Exhibit K



DHHS Information Security Requirements

- e. limit disclosure of the Confidential Information to the extent permitted by law.
- f. Confidential Information received under this Contract and individually identifiable data derived from DHHS Data, must be stored in an area that is physically and technologically secure from access by unauthorized persons during duty hours as well as non-duty hours (e.g., door locks, card keys, biometric identifiers, etc.).
- g. only authorized End Users may transmit the Confidential Data, including any derivative files containing personally identifiable information, and in all cases, such data must be encrypted at all times when in transit, at rest, or when stored on portable media as required in section IV above.
- h. in all other instances Confidential Data must be maintained, used and disclosed using appropriate safeguards, as determined by a risk-based assessment of the circumstances involved.
- i. understand that their user credentials (user name and password) must not be shared with anyone. End Users will keep their credential information secure. This applies to credentials used to access the site directly or indirectly through a third party application.

Contractor is responsible for oversight and compliance of their End Users. DHHS reserves the right to conduct onsite inspections to monitor compliance with this Contract, including the privacy and security requirements provided in herein, HIPAA, and other applicable laws and Federal regulations until such time the Confidential Data is disposed of in accordance with this Contract.

V. LOSS REPORTING

The Contractor must notify the State's Privacy Officer and Security Officer of any Security Incidents and Breaches immediately, at the email addresses provided in Section VI.

The Contractor must further handle and report Incidents and Breaches involving PHI in accordance with the agency's documented Incident Handling and Breach Notification procedures and in accordance with 42 C.F.R. §§ 431.300 - 306. In addition to, and notwithstanding, Contractor's compliance with all applicable obligations and procedures, Contractor's procedures must also address how the Contractor will:

- 1. Identify Incidents;
- 2. Determine if personally identifiable information is involved in Incidents;
- 3. Report suspected or confirmed Incidents as required in this Exhibit or P-37;
- 4. Identify and convene a core response group to determine the risk level of Incidents and determine risk-based responses to Incidents; and

Contractor Initials _____

V5. Last update 10/09/18

Exhibit K
DHHS Information
Security Requirements
Page 8 of 9

Date 3/11/2022

New Hampshire Department of Health and Human Services Exhibit K



DHHS Information Security Requirements

 Determine whether Breach notification is required, and, if so, identify appropriate Breach notification methods, timing, source, and contents from among different options, and bear costs associated with the Breach notice as well as any mitigation measures.

Incidents and/or Breaches that implicate PI must be addressed and reported, as applicable, in accordance with NH RSA 359-C:20.

VI. PERSONS TO CONTACT

A. DHHS Privacy Officer:

DHHSPrivacyOfficer@dhhs.nh.gov

B. DHHS Security Officer:

DHHSInformationSecurityOffice@dhhs.nh.gov

Contractor Initials _____

V5. Last update 10/09/18

Exhibit K
DHHS Information
Security Requirements
Page 9 of 9

Date ____

State of New Hampshire Department of State

CERTIFICATE

I, William M. Gardner, Secretary of State of the State of New Hampshire, do hereby certify that BRETTON HEALTHCARE SOLUTIONS LLC is a New Hampshire Limited Liability Company registered to transact business in New Hampshire on October 13, 2020. I further certify that all fees and documents required by the Secretary of State's office have been received and is in good standing as far as this office is concerned.

Business ID: 853792

Certificate Number: 0005703493



IN TESTIMONY WHEREOF,

I hereto set my hand and cause to be affixed the Seal of the State of New Hampshire, this 14th day of March A.D. 2022.

William M. Gardner

Secretary of State

CERTIFICATE OF VOTE/AUTHORITY

١.	G	reg Carbone of theBretton Healthcare Solutions
loon II	iiiic	d Liability Company under RSA 304-C, do hereby certify that:
	1.	I am the (managing member, etc) of theBretton Healthcare Solutions, LLC:
		This Limited Liability Company may enter into any and all contracts, amendments, renewals, revisions or modifications thereto, with the State of New Hampshire, acting through its Department of Health and Human Services.
	24	RESOLVED: That the (Managing Member) is hereby authorized on behalf of this company to enter into said contracts with the State, and to execute any and all documents, agreements, and other instruments, and any amendments, revisions, or modifications thereto, as he/she may deem necessary, desirable or appropriate, and (Greg Carbone) is the duly elected (Manager Member) of the Limited Liability Company.
	2.	I further certify that it is understood that the State of New Hampshire will rely on this certificate as evidence that the person listed above currently occupies the position indicated and that they have full authority to bind the LLC and that this authorization shall remain valid for thirty (30) days from the date of this certificate.
		Name: Greg Carbone Title: Principal and Managing Member Company Name: Bretton Healthcare Solutions
		つ/ 友 / クベ

BRETTON HEALTHCARE SOLUTIONS LLC

Operating Agreement

This Operating Agreement of Bretton Healthcare Solutions LLC (the "Company") is made as of October 13, 2020, by and between the persons identified as the Managers and Members on Schedule A attached hereto (such persons and their respective successors in office or in interest being hereinafter referred to individually as a "Manager" or "Member" or collectively as the "Managers" or "Members").

WHEREAS, the Company was formed as a limited liability company under the New Hampshire Limited Liability Company Act, RSA 304-C (as amended from time to time, the "Act") on October 13, 2020; and

WHEREAS, the Managers and the Members wish to set out fully their respective rights, obligations and duties regarding the Company and its assets and liabilities;

NOW, THEREFORE, in consideration of the mutual covenants expressed herein, the parties hereby agree as follows:

ARTICLE I - Organization and Powers

- 1.1. Organization. The Company has been formed by the filing of its Certificate of Formation with the New Hampshire Secretary of State pursuant to the Act. The Certificate of Formation may be restated by the Managers as provided in the Act or amended by the Managers to change the address of the office of the Company in New Hampshire and the name and address of its resident agent in New Hampshire or to make corrections required by the Act. Other additions to or amendments of the Certificate of Formation shall be authorized by the Members as provided in Section 2.5. The Certificate of Formation, as so amended from time to time, is referred to herein as the "Certificate." The Managers shall deliver a copy of the Certificate and any amendment thereto to any Member who so requests.
- 1.2. <u>Purposes and Powers</u>. The principal business activity and purpose of the Company shall initially be to engage in medical staffing and consulting and to engage in any business related thereto or useful in connection therewith. However, the business and purposes of the Company shall not be limited to its initial principal business activity and, unless the Members otherwise determine, the Company shall have authority to engage in any other lawful business, trade, purpose or activity permitted by the Act, and it shall possess and may exercise all of the powers and privileges granted by the Act and any powers incidental thereto, so far as such powers and privileges are necessary or convenient to the conduct, promotion or attainment of the business, purposes or activities of the Company, including without limitation the following powers:
- (a) to conduct its business and operations in any state, territory or possession of the United States or in any foreign country or jurisdiction;

- (b) to purchase, receive, take, lease or otherwise acquire, own, hold, improve, maintain, use or otherwise deal in and with, sell, convey, lease, exchange, transfer or otherwise dispose of, mortgage, pledge, encumber or create a security interest in all or any of its real or personal property, or any interest therein, wherever situated;
- (c) to borrow or lend money or obtain or extend credit and other financial accommodations, to invest and reinvest its funds in any type of security or obligation of or interest in any public, private or governmental entity, and to give and receive interests in real and personal property as security for the payment of funds so borrowed, loaned or invested;
- (d) to make contracts, including contracts of insurance, incur liabilities and give guaranties, whether or not such guaranties are in furtherance of the business and purposes of the Company, including without limitation guaranties of obligations of other persons who are interested in the Company or in whom the Company has an interest, and to grant interests in real and personal property as security for such contracts, liabilities and guaranties;
- (e) to appoint one or more Managers of the Company, to employ officers, employees, agents and other persons, to fix the compensation and define the duties and obligations of such personnel, to establish and carry out retirement, incentive and benefit plans for such personnel and to indemnify such personnel to the extent permitted by this Agreement and the Act;
- (f) to make donations irrespective of benefit to the Company for the public welfare or for community, charitable, religious, educational, scientific, civic or similar purposes;
- (g) to institute, prosecute and defend any legal action or arbitration proceeding involving the Company, and to pay, adjust, compromise, settle or refer to arbitration any claim by or against the Company or any of its assets;
 - (h) to act as the voting trustee in one or more voting trusts;
 - (i) to be a partner in one or more partnerships;
- (j) to enter into or engage in any kind of activity necessary to, in connection with, or incidental to, the accomplishment of the purposes of the Company, so long as said activities may be lawfully carried on or performed by a limited liability Company under the laws of the State of New Hampshire; and
- (k) to take any other action not prohibited under the Act or other applicable law.
- 1.3. <u>Principal Place of Business</u>. The principal office and place of business of the Company and the place where the books and records of the Company shall be maintained shall be at 84 Branch Tumpike, Unit #118, Concord, NH 03301. After giving notice to the Members, the Managers may change the principal office or place of business of the Company at any time and may cause the Company to establish other offices or places of business.

- 1.4. <u>Fiscal Year</u>. The fiscal year of the Company shall end on December 31 in each year.
- 1.5. Qualification in Other Jurisdictions. The Managers shall cause the Company to be qualified or registered under applicable laws of any jurisdiction in which the Company transacts business and shall be authorized to execute, deliver and file any certificates and documents necessary to effect such qualification or registration, including without limitation the appointment of agents for service of process in such jurisdictions.

ARTICLE II - Members

- 2.1. <u>Members</u>. The initial Members of the Company and their addresses shall be listed on <u>Schedule A</u> and such Schedule shall be amended from time to time by the Managers to reflect the withdrawal of Members or the admission of new or additional Members pursuant to this Agreement. <u>Schedule A</u> shall set forth the percentage interest which each Member holds in the profits and losses of the Company (the "<u>Membership Interests</u>"), and the Contribution (as defined in Section 6.2) of each Member. The Members shall constitute a single class or group of Members of the Company for all purposes of the Act, unless otherwise explicitly provided herein. The Managers shall notify the Members of changes in <u>Schedule A</u>, which shall constitute the record list of the Members for all purposes of this Agreement.
- 2.2. <u>Admission of New Members</u>. Additional persons may be admitted to the Company as Members and may participate in the profits, losses, distributions, allocations and capital contributions of the Company upon such terms as are established by the Managers, which may include the establishment of classes or groups of one or more Members having different relative rights, powers and duties, or the right to vote as a separate class or group on specified matters, by amendment of this Agreement under Section 10.4. Existing Members shall have no preemptive or similar right to subscribe to the purchase of new Membership Interests in the Company.

2.3. Meetings of Members.

(a) Meetings of Members may be called (i) for any proper purpose at any time by the Managers or the holders of a majority of the Membership Interests, and (ii) at any time when the management of the Company is vested in fewer than all of the Members, by any Member for the purpose of considering the affairs of the Company and taking any action permitted to be taken by the Members by law, this Agreement or the Certificate (unless there has been a meeting pursuant to this clause (ii) within 240 days prior to the delivery of the demand). The Managers or the Members calling a meeting pursuant to clause (i) above shall determine the date, time and place of such meeting of Members, and written notice thereof shall be given by the Managers to each Member not less than seven days or more than 60 days prior to the date of the meeting. The Managers shall determine the date of any meeting called pursuant to clause (ii) above (which shall not be more than 30 days after the date of the applicable Member's written demand therefor), time and place of each meeting of Members, and written notice thereof shall be given by the Managers to each Member not less than seven days or more than 30 days prior to the date of the meeting. Notice shall be sent to Members of record on the date when the meeting

is called. The business of each meeting of Members shall be limited to the purposes described in the notice. A written waiver of notice, executed before or after a meeting by a Member or its authorized attorney and delivered to the Managers, shall be deemed equivalent to notice of the meeting.

- (b) A chairman selected by the Managers shall preside at all meetings of the Members unless the Members elect from the membership a chairman of the meeting. The chairman shall determine the order of business and the procedures to be followed at each meeting of Members.
- (c) Members (i) holding a majority of the Membership Interests and (ii), in the case of a meeting of Members pursuant to Section 2.3(a)(ii) only, constituting more than one-half (1/2) in number of the Members, shall constitute a quorum for the transaction of any business at a meeting of Members. Members may attend a meeting in person or by proxy. Members may also participate in a meeting by means of conference telephone or similar communications equipment that permits all Members present to hear each other. If less than a quorum of the Members is present, the meeting may be adjourned by the chairman to a later date, time and place, and the meeting may be held as adjourned without further notice. When an adjourned meeting is reconvened, any business may be transacted that might have been transacted at the original meeting.
- (d) At any meeting of the Members pursuant to Section 2.3(a)(ii), the Managers shall report on the affairs of the Company and shall either make available in writing the information listed in Sections 28, I(a) (e) of the Act or be prepared to respond promptly in writing to demands for information pursuant to Section 28 of the Act.
- 2.4. Action Without a Meeting. There is no requirement that the Members hold a meeting in order to take action on any matter. Any action required or permitted to be taken by the Members may be taken without a meeting if one or more written consents to such action shall be signed by Members who hold the Membership Interests or other interest in the Company required to approve the action being taken. Such written consents shall be delivered to the Managers at the principal office of the Company and unless otherwise specified shall be effective on the date when the first consent is so delivered. The Managers shall give prompt notice to all Members who did not consent to any action taken by written consent of Members without a meeting.
- 2.5. <u>Voting Rights</u>. Unless otherwise required by the Act or this Agreement, all actions, approvals and consents to be taken or given by the Members under the Act, this Agreement or otherwise shall require the affirmative vote or written consent of Members holding a majority of the Membership Interests.
- 2.6. <u>Limitation of Liability of Members</u>. Except as otherwise provided in the Act, no Member of the Company shall be obligated personally for any debt, obligation or liability of the Company or of any other Member, whether arising in contract, tort or otherwise, solely by reason of being a Member of the Company. Except as otherwise provided in the Act, by law or expressly in this Agreement, no Member shall have any fiduciary or other duty to another

Member with respect to the business and affairs of the Company, and no Member shall be liable to the Company or any other Member for acting in good faith reliance upon the provisions of this Agreement. No Member shall have any responsibility to restore any negative balance in its Capital Account (as defined in Section 6.1) or to contribute to or in respect of the liabilities or obligations of the Company or return distributions made by the Company except as required by the Act or other applicable law; provided, however, that Members are responsible for their failure to make required Contributions under Section 6.2. The failure of the Company to observe any formalities or requirements relating to the exercise of its powers or the management of its business or affairs under this Agreement or the Act shall not be grounds for making its Members or Managers responsible for the liabilities of the Company.

- 2.7. <u>Authority</u>. Unless specifically authorized by the Managers, no Member that is not a Manager shall be an agent of the Company or have any right, power or authority to act for or to bind the Company or to undertake or assume any obligation or responsibility of the Company or of any other Member.
- 2.8. No Right to Withdraw. No Member shall have any right to resign or withdraw from the Company without the consent of the other Members or to receive any distribution or the repayment of its capital contribution except as provided in Section 7.2 and Article IX upon dissolution and liquidation of the Company. No Member shall have any right to have the fair value of its Membership Interest in the Company appraised and paid out upon the resignation or withdrawal of such Member or any other circumstances.
- 2.9. <u>Rights to Information</u>. Members shall have the right to receive from the Managers upon request a copy of the Certificate and of this Agreement, as amended from time to time, and such other information regarding the Company as is required by the Act, subject to reasonable conditions and standards established by the Managers, as permitted by the Act, which may include without limitation withholding or restricting the use of confidential information.

ARTICLE III - Management

- 3.1. <u>Managers</u>. Gregory B. Carbone shall be the initial Manager(s) of the Company. The names and addresses of the Manager(s) shall be listed on <u>Schedule A</u> and such Schedule shall be amended from time to time by the Manager(s) to reflect the resignation or removal of Manager(s) or the appointment of new or additional Manager(s) pursuant to this Agreement.
- 3.2. Qualification. Each Manager shall devote such time to the business and affairs of the Company as is reasonably necessary for the performance of such Manager's duties, but shall not be required to devote full time to the performance of such duties and may delegate its responsibilities as provided in Section 3.3.
- 3.3. <u>Powers and Duties of the Manager(s)</u>. The business and affairs of the Company shall be managed under the direction of the Manager(s), who shall have and may exercise on behalf of the Company all of its rights, powers, duties and responsibilities under Section 1.2 or as provided by law, including without limitation the right and authority:

- (a) to manage the business and affairs of the Company and for this purpose to employ, retain or appoint any officers, employees, consultants, agents, brokers, professionals or other persons in any capacity for such compensation and on such terms as the Manager(s) deem necessary or desirable and to delegate to such persons such of their duties and responsibilities as the Manager(s) shall determine;
- (b) to enter into, execute, deliver, acknowledge, make, modify, supplement or amend any documents or instruments in the name of the Company;
- (c) to borrow money or otherwise obtain credit and other financial accommodations on behalf of the Company on a secured or unsecured basis as provided in Section 1.2(c), and to perform or cause to be performed all of the Company's obligations in respect of its indebtedness and any mortgage, lien or security interest securing such indebtedness; and
- (d) to make elections and prepare and file returns regarding any federal, state or local tax obligations of the Company.

Unless otherwise provided in this Agreement, any action taken by a Manager, and the signature of a Manager on any agreement, contract, instrument or other document on behalf of the Company, shall be sufficient to bind the Company and shall conclusively evidence the authority of that Manager and the Company with respect thereto.

- 3.4. <u>Tax Matters Partner</u>. The Member so designated by the Manager(s) from time to time shall serve as the "Tax Matters Partner" of the Company for purposes of Section 6231(a)(7) of the Internal Revenue Code of 1986 as amended (the "<u>Code</u>"), with power to manage and represent the Company in any administrative proceeding of the Internal Revenue Service. The initial Tax Matters Partner of the Company shall be Gregory B. Carbone.
- 3.5. Reliance by Third Parties. Any person dealing with the Company, the Manager(s) or any Member may rely upon a certificate signed by any Manager as to (a) the identity of any Manager or Member; (b) any factual matters relevant to the affairs of the Company; (c) the persons who are authorized to execute and deliver any document on behalf of the Company; or (d) any action taken or omitted by the Company, the Manager(s) or any Member.
- 3.6. <u>Resignation and Removal</u>. Any Manager may resign upon at least 60 days notice to the Members and the other Manager(s) (unless notice is waived by them). Any Manager may be removed at any time with or without cause by the unanimous affirmative vote or written consent of the Members.
- 3.7. <u>Appointment of New or Additional Manager(s)</u>. New or additional Managers may be appointed at any time by the unanimous affirmative vote or written consent of the Members.
- 3.8. <u>Meetings and Action of Manager(s)</u>. Unless otherwise determined by the Members or Manager(s), all action to be taken by the Manager(s) shall be taken by majority vote

or written consent of a majority of the Managers then in office. There is no requirement that the Manager(s) hold a meeting in order to take action on any matter. Meetings of the Managers may be called by any Manager. If action is to be taken at a meeting of the Manager(s), notice of the time, date and place of the meeting shall be given to each Manager by an officer or the Manager calling the meeting by personal delivery, telephone or fax sent to the business or home address of each Manager at least 24 hours in advance of the meeting, or by written notice mailed to each Manager at either such address at least 72 hours in advance of the meeting; however, no notice need be given to a Manager who waives notice before or after the meeting, or who attends the meeting without protesting at or before its commencement the inadequacy of notice to him or her. Managers may also attend a meeting in person or by proxy, and they may also participate in a meeting by means of conference telephone or similar communications equipment that permits all Managers present to hear each other. A chairman selected by the Managers shall preside at all meetings of the Managers. The chairman shall determine the order of business and the procedures to be followed at each meeting of the Managers.

3.9. <u>Limitation of Liability of Manager or Member</u>. No Manager or Member shall be obligated personally for any debt, obligation or liability of the Company or of any Member, whether arising in contract, tort or otherwise, solely by reason of being or acting as Manager of the Company. No Manager or Member shall be personally liable to the Company or to its Members for breach of any fiduciary or other duty that does not involve (a) a breach of the duty of loyalty to the Company or its Members, (b) acts or omissions not in good faith or which involve gross negligence, intentional misconduct or a knowing violation of law; or (c) a transaction from which the Manager or Member derived an improper personal benefit.

ARTICLE IV - Indemnification

4.1. <u>Definitions</u>. For purposes of this Article:

"Manager" includes (a) a person serving as a Manager or an officer of the Company or in a similar executive capacity appointed by the Manager(s) and exercising rights and duties delegated by the Manager(s), (b) a person serving at the request of the Company as a director, Manager, officer, employee or other agent of another organization, and (c) any person who formerly served in any of the foregoing capacities;

"expenses" means all expenses, including attorneys' fees and disbursements, actually and reasonably incurred in defense of a proceeding or in seeking indemnification under this Article, and except for proceedings by or in the right of the Company or alleging that a Manager received an improper personal benefit, any judgments, awards, fines, penalties and reasonable amounts paid in settlement of a proceeding; and

"proceeding" means any threatened, pending or completed action, suit or proceeding, whether civil, criminal, administrative or investigative, and any claim which could be the subject of a proceeding.

4.2. <u>Right to Indemnification</u>. Except as limited by law and subject to the provisions of this Article, the Company shall indemnify each of its Managers against all expenses incurred

by them in connection with any proceeding in which a Manager is involved as a result of serving in such capacity, except that no indemnification shall be provided for a Manager regarding any matter as to which it shall be finally determined that such Manager did not act in good faith and in the reasonable belief that its action was in the best interests of the Company. Subject to the foregoing limitations, such indemnification may be provided by the Company with respect to a proceeding in which it is claimed that a Manager received an improper personal benefit by reason of its position, regardless of whether the claim arises out of the Manager's service in such capacity, except for matters as to which it is finally determined that an improper personal benefit was received by the Manager.

- 4.3. Award of Indemnification. The determination of whether the Company is authorized to indemnify a Manager hereunder and any award of indemnification shall be made in each instance (a) by a majority of the Managers who are not parties to the proceeding in question, (b) by independent legal counsel appointed by the Manager(s) or the Members or (c) by the holders of a majority of the Membership Interests of the Members who are not parties to the proceeding in question. The Company shall be obliged to pay indemnification applied for by a Manager unless there is an adverse determination (as provided above) within forty-five (45) days after the application. If indemnification is denied, the applicant may seek an independent determination of its right to indemnification by a court, and in such event, the Company shall have the burden of proving that the applicant was ineligible for indemnification under this Article. Notwithstanding the foregoing, in the case of a proceeding by or in the right of the Company in which a Manager is adjudged liable to the Company, indemnification hereunder shall be provided to such Manager only upon a determination by a court having jurisdiction that in view of all the circumstances of the case, such Manager is fairly and reasonably entitled to indemnification for such expenses as the court shall deem proper.
- 4.4. Successful Defense. Notwithstanding any contrary provisions of this Article, if a Manager has been wholly successful on the merits in the defense of any proceeding in which it was involved by reason of its position as Manager or as a result of serving in such capacity (including termination of investigative or other proceedings without a finding of fault on the part of the Manager), the Manager shall be indemnified by the Company against all expenses incurred by the Manager in connection therewith.
- 4.5. Advance Payments. Except as limited by law, expenses incurred by a Manager in defending any proceeding, including a proceeding by or in the right of the Company, shall be paid by the Company to the Manager in advance of final disposition of the proceeding upon receipt of its written undertaking to repay such amount if the Manager is determined pursuant to this Article or adjudicated to be ineligible for indemnification, which undertaking shall be an unlimited general obligation but need not be secured and may be accepted without regard to the financial ability of the Manager to make repayment; provided, however, that no such advance payment of expenses shall be made if it is determined pursuant to Section 4.3 of this Article on the basis of the circumstances known at the time (without further investigation) that the Manager is ineligible for indemnification.
- 4.6. <u>Insurance</u>. The Company shall have power to purchase and maintain insurance on behalf of any Manager, officer, agent or employee against any liability or cost incurred by such

person in any such capacity or arising out of its status as such, whether or not the Company would have power to indemnify against such liability or cost.

- 4.7. <u>Heirs and Personal Representatives</u>. The indemnification provided by this Article shall inure to the benefit of the heirs and personal representatives of each Manager.
- 4.8. <u>Non-Exclusivity</u>. The provisions of this Article shall not be construed to limit the power of the Company to indemnify its Managers, Members, officers, employees or agents to the full extent permitted by law or to enter into specific agreements, commitments or arrangements for indemnification permitted by law. The absence of any express provision for indemnification herein shall not limit any right of indemnification existing independently of this Article.
- 4.9. <u>Amendment</u>. The provisions of this Article may be amended or repealed in accordance with Section 10.5; however, no amendment or repeal of such provisions that adversely affects the rights of a Manager under this Article with respect to its acts or omissions at any time prior to such amendment or repeal shall apply to such Manager without its consent.

ARTICLE V - Conflicts of Interest

- 5.1. <u>Transactions with Interested Persons</u>. Unless entered into in bad faith, no contract or transaction between the Company and one or more of its Managers or Members, or between the Company and any other corporation, partnership, association or other organization in which one or more of its Managers or Members have a financial interest or are directors, partners, Managers or officers, shall be voidable solely for this reason or solely because such Manager or Member was present or participated in the authorization of such contract or transaction if:
- (a) the material facts as to the relationship or interest of such Manager or Member and as to the contract or transaction were disclosed or known to the other Managers (if any) or Members and the contract or transaction was authorized by the disinterested Managers (if any) or Members; or
- (b) the contract or transaction was fair to the Company as of the time it was authorized, approved or ratified by the disinterested Managers (if any) or Members;

and no Manager or Member interested in such contract or transaction, because of such interest, shall be considered to be in breach of this Agreement or liable to the Company, any Manager or Member, or any other person or organization for any loss or expense incurred by reason of such contract or transaction or shall be accountable for any gain or profit realized from such contract or transaction.

5.2. Other Business Ventures and Investment Opportunities. The Members and Manager(s) may engage in and possess interests in other business ventures and investment opportunities of every kind and description, independently or with others, including serving as directors, managers and general partners of other corporations, limited liability companies and partnerships with purposes similar to those of the Company. Neither the Company nor any other

Member shall have any rights in or to such ventures or opportunities or the income or profits therefrom.

ARTICLE VI - Capital Accounts and Contributions

6.1. Capital Accounts.

- (a) There shall be established on the books of the Company a separate capital account (a "Capital Account") for each Member.
- (b) The Capital Account of each Member (regardless of the time or manner in which such Member's interest was acquired) shall be maintained in accordance with the rules of Section 704(b) of the Internal Revenue Code of 1986, as amended, from time to time (the "Code"), and Treasury Regulation Section 1.704-1(b)(2)(iv). Adjustments shall be made to the Capital Accounts for distributions and allocations as required by the rules of Section 704(b) of the Code and the Treasury Regulations thereunder.
- (c) If there is a transfer of all or a part of an interest in the Company by a Member, the Capital Account of the transferor that is attributable to the transferred interest shall carry over to the transferee of such Member.
- (d) Subject to Section 7.2, notwithstanding any other provision contained herein to the contrary, no Member shall be required to restore any negative balance in its Capital Account.
- 6.2. Contributions. Each Member shall make the contributions to the capital of the Company (herein "Contributions") specified on Schedule A. All Contributions shall be paid in cash unless otherwise specified on Schedule A or agreed to by the Members. Except as set forth on Schedule A, no Member or Manager shall be entitled or required to make any contribution to the capital of the Company; however, the Company may borrow from its Members as well as from banks or other lending institutions to finance its working capital or the acquisition of assets upon such terms and conditions as shall be approved by the Manager(s), and any such borrowing from Members shall not be considered Contributions or reflected in their Capital Accounts. The value of all non-cash Contributions made by Members shall be set forth on Schedule A. No Member shall be entitled to any interest or compensation with respect to its Contribution or any services rendered on behalf of the Company except as specifically provided in this Agreement or approved by the Manager(s). No Member shall have any liability for the repayment of the Contribution of any other Member and each Member shall look only to the assets of the Company for return of its Contribution.

ARTICLE VII - Profits, Losses and Distributions

7.1. Profits, Losses and Distributions.

- (a) All profits and losses arising from the normal course of business operations or otherwise and all cash available for distribution from whatever source, commencing with the date of this Agreement, shall be allocated or distributed to the Members according to their Membership Interests.
- (b) All profits and losses allocated to the Members shall be credited or charged, as the case may be, to their Capital Accounts. The terms "profits" and "losses" as used in this Agreement shall mean income and losses, and each item of income, gain, loss, deduction or credit entering into the computation thereof, as determined in accordance with the accounting methods followed by the Company and computed in a manner consistent with Treasury Regulation Section 1.704-1(b)(2)(iv). Profits and losses for Federal income tax purposes shall be allocated in the same manner as profits and losses for purposes of this Article VII, except as provided in Section 7.3(a).

7.2. <u>Distributions Upon Dissolution</u>.

- (a) Upon dissolution and termination, after payment of, or adequate provision for, the debts and obligations of the Company, the remaining assets of the Company (or the proceeds of sales or other dispositions in liquidation of the Company's assets, as may be determined by the Member(s)) shall be distributed to the Members in accordance with the positive balances in their Capital Accounts after taking into account all Capital Account adjustments for the Company taxable year in which the liquidation occurs.
- With respect to assets distributed in kind to the Members in liquidation or otherwise, (i) any unrealized appreciation or unrealized depreciation in the values of such assets shall be deemed to be profits and losses realized by the Company immediately prior to the liquidation or other distribution event; and (ii) such profits and losses shall be allocated to the Members and credited or charged to their Capital Accounts, and any property so distributed shall be treated as a distribution of an amount in cash equal to the excess of such fair market value over the outstanding principal balance of and accrued interest on any debt by which the property is encumbered. For the purposes of this Section 7.2(b), "unrealized appreciation" or "unrealized depreciation" shall mean the difference between the fair market value of such assets, taking into account the fair market value of the associated financing but subject to Section 7701(g) of the Code, and the Company's basis in such assets as determined under Treasury Regulation Section 1.704-1(b). This Section 7.2(b) is merely intended to provide a rule for allocating unrealized gains and losses upon liquidation or other distribution event, and nothing contained in this Section 7.2(b) or elsewhere in this Agreement is intended to treat or cause such distributions to be treated as sales for value. The fair market value of such assets shall be determined by an appraiser to be selected by the Manager with the consent of the Members.

7.3. Special Provisions.

Notwithstanding the foregoing provisions in this Article VII:

- (a) Income, gain, loss and deduction with respect to Company property which has a variation between its basis computed in accordance with Treasury Regulation Section 1.704-(b) and its basis computed for Federal income tax purposes shall be shared among Members so as to take account of the variation in a manner consistent with the principles of Section 704(c) of the Code and Treasury Regulation Section 1.704-3.
- (b) Section 704 of the Code and the Treasury Regulations issued thereunder, including but not limited to the provisions of such regulations addressing qualified income offset provisions, minimum gain chargeback requirements and allocations of deductions attributable to nonrecourse debt and partner nonrecourse debt, are hereby incorporated by reference into this Agreement.
- 7.4. <u>Distribution of Assets in Kind</u>. No Member shall have the right to require any distribution of any assets of the Company to be made in cash or in kind. If the Manager(s) determine to distribute assets of the Company in kind, such assets shall be distributed on the basis of their fair market value as determined by the Manager(s). Any Member entitled to any interest in such assets shall, unless otherwise determined by the Manager(s), receive separate assets of the Company, and not an interest as tenant-in-common with other Members so entitled in each asset being distributed. Distributions in kind need not be made on a pro-rata basis but may be made on any basis which the Manager(s) determine to be reasonable under the circumstances.

ARTICLE VIII - Transfers of Interests

8.1. Transfer of a Member's Membership Interest.

- (a) Except as set forth in the first sentence of Section 8.2, no Member may sell, assign, give, pledge, hypothecate, encumber or otherwise transfer, including, without limitation, any assignment or transfer by operation of law or by order of court, such Member's Membership Interest in the Company or any part thereof, or in all or any part of the assets of the Company, without the prior written consent of a majority of the Managers and in accordance with the terms of Section 8.1(b). The granting or denying of such consent shall be in the Managers' absolute discretion. The Company shall not be required to recognize any such assignment until the instrument conveying such Membership Interest has been delivered to one of the Managers for recording on Schedule A hereto. Any attempted sale, transfer, assignment, pledge or other disposition in contravention of the provisions of this section shall be void and ineffectual and shall not bind, or be recognized, by the Company.
- (b) After obtaining the prior written consent of a majority of the Managers, but before any Membership Interest or any part thereof may be sold, assigned, gifted, pledged, hypothecated, encumbered or otherwise transferred, including transfer by operation of law or by order of court, the Member holding such Membership Interest proposing such sale or transfer (the "transferor") shall first give written notice thereof to the other Members at least sixty (60) days prior to the proposed date of transfer stating the proposed transferee, the Membership

Interest proposed to be transferred, the purchase price, if any, and the terms of the proposed transaction. Remaining Members receiving such notice (the "Purchasing Members") shall thereupon have the option, but not the obligation, to acquire all, but not less than all, of the Membership Interest proposed to be sold or transferred for the Purchase Price determined pursuant to Section 8.1(d) (the "Purchase Price"). Within thirty (30) days after the giving of such notice by the transferor, each Purchasing Member shall give written notice to the transferor stating whether or not he or she elects to exercise the option to purchase and a date and time (the "Closing Date") for the consummation of the purchase not less than sixty (60) or more than ninety (90) days after the giving of such notice. If two (2) or more Purchasing Members desire to purchase the Membership Interest proposed to be sold or transferred, then, in the absence of an agreement between or among them, each such Purchasing Member shall purchase the Membership Interest proposed to be sold or transferred in the proportion that his or her Membership Interest bears to the total Membership Interests of all the Purchasing Members who desire to so purchase. Failure by a Purchasing Member to give such notice within such time period shall be deemed an election by such Purchasing Member not to exercise such option. If the Purchase Price is determined by appraisal as set forth in Section 8.1(d)(ii), a Purchasing Member may rescind his or her election to purchase by written notice to the transferor given within ten (10) days after being notified of the determination of the appraisers.

- (c) If the Membership Interest offered hereunder is not purchased within the time period stated above, the transferor may transfer such Membership Interest at any time during the 60-day period after the termination of such time period, but only upon the terms and to the transferee stated in its notice delivered pursuant to subsection (b). After such Membership Interest is so transferred, or if the transfer is not consummated within such period, the Membership Interest shall again become subject to the terms of this Agreement.
 - (d) The Purchase Price shall be determined as follows:
 - (i) In the case of a proposed sale or transfer under paragraph (b) to a third party in a bona fide transaction for fair value payable in cash or the equivalent currently or in future installments, the Purchase Price for such Membership Interests shall be the value offered by such third party payable upon the same terms.
 - (ii) In all other cases, including without limitation a proposed transfer or other disposition not constituting a sale described in subsection (i), the Purchase Price shall be the fair market value of the Membership Interest being purchased as of the last day of the month immediately prior to the month during which the transferor gave his or her notice. "Fair market value" as of any date shall mean the cash price obtainable in an arm'slength sale between an informed and willing buyer (under no compulsion to purchase) and an informed and willing seller (under no compulsion to sell) of the Membership Interest, based upon the going concern value of the Company, taking into account any minority or non-control discount. If the parties are unable to agree upon the fair market value, such fair market value shall be determined by appraisal as follows: Either party may

require appraisal by giving written notice to the other party and appointing an independent appraiser. The other party shall deliver a written notice appointing an independent appraiser within fifteen (15) days after receipt of the notice from the other. The two appraisers so appointed, or if only one appraiser is appointed, that appraiser, shall promptly seek to determine the fair market value. If the two appraisers cannot agree within thirty (30) days of their appointment, a third independent appraiser shall be chosen within ten (10) days thereafter by the mutual consent of such first two appraisers or, if such first two appraisers fail to agree upon the appointment of a third appraiser, such appointment shall be made by the Boston office of the American Arbitration Association, or any organization successor thereto, and shall be a disinterested person qualified in the valuation of the Company's business. The three appraisers shall make the determination in accordance with the rules of the American Arbitration Association or any such successor then in effect, and such determination shall be binding and conclusive on the parties. Each party shall pay the costs of its own appraiser and shall share equally in the costs, if any, of a third appraiser and any other costs of arbitration, excluding their own costs.

8.2. Death or Incompetence of a Member. If a Member dies, such Member's executor, administrator, or trustee, or, if he or she is adjudicated incompetent, such Member's guardian, or, if it is a corporation, trust, limited liability company or partnership and is dissolved, the liquidator, shall automatically become an assignee (the "Assignee") of the Membership Interest of the deceased, incompetent, or dissolved Member. The Assignee may receive distributions and shall have all the rights of a Member for the purpose of settling or managing such deceased or incompetent Member's estate, but shall not be a Member and shall not have the power to vote such Member's Membership Interest. The Assignee shall also have such power as the decedent, incompetent or dissolved entity possessed to: (a) assign all or any part of the Member's Membership Interest subject to Section 8.1; and (b) to satisfy conditions precedent to the assignment of the Membership Interest set forth in Section 8.1.

8.3. Admission of Member: Effect of Transfer.

- (a) In no event may any person obtaining a Membership Interest in the Company by assignment, transfer, pledge or other means from an existing Member be admitted as a successor Member without the affirmative vote or written consent of Members exclusive, in each case, of the Member whose Membership Interest is being transferred.
- (b) If the transferee is admitted as a Member or is already a Member, the Member transferring its Membership Interest shall be relieved of liability with respect to the transferred Membership Interest arising or accruing under this Agreement on or after the effective date of the transfer, unless the transferor affirmatively assumes such liability; provided, however, that the transferor shall not be relieved of any liability for prior distributions and unpaid contributions unless the transferee affirmatively assumes such liabilities.

(c) Any person who acquires in any manner a Membership Interest or any part thereof in the Company, whether or not such person has accepted and assumed in writing the terms and provisions of this Agreement or been admitted as a Member, shall be deemed by the acquisition of such Membership Interest to have agreed to be subject to and bound by all of the provisions of this Agreement with respect to such Membership Interest, including without limitation, the provisions hereof with respect to any subsequent transfer of such Membership Interest.

ARTICLE IX - Dissolution, Liquidation and Termination

- 9.1. <u>Dissolution</u>. The Company shall dissolve and its affairs shall be wound up upon the first to occur of the following:
 - (a) the written consent of the Members;
 - (b) the entry of a decree of judicial dissolution under Section 51 of the Act; or
- (c) The consolidation or merger of the Company in which it is not the resulting or surviving entity.
- 9.2. <u>Liquidation</u>. Upon dissolution of the Company, the Managers shall act as its liquidating trustees or the Managers may appoint one or more Managers or Members as liquidating trustee. The liquidating trustees shall proceed diligently to liquidate the Company and wind up its affairs and shall dispose of the assets of the Company as provided in Section 7.2. Until final distribution, the liquidating trustees may continue to operate the business and properties of the Company with all of the power and authority of the Managers. As promptly as possible after dissolution and again after final liquidation, the liquidating trustees shall cause an accounting by the accounting firm then serving the Company of the Company's assets, liabilities, operations and liquidating distributions to be given to the Members.
- 9.3. <u>Certificate of Cancellation</u>. Upon completion of the distribution of Company assets as provided herein, the Company shall be terminated, and the Managers (or such other person or persons as the Act may require or permit) shall file a Certificate of Cancellation with the Secretary of State of New Hampshire under the Act, cancel any other filings made pursuant to Sections 1.1, 1.3 and 1.5 and take such other actions as may be necessary to terminate the existence of the Company.

ARTICLE X - General Provisions

- 10.1. Offset. Whenever the Company is obligated to make a distribution or payment to any Member, any amounts that Member owes the Company may be deducted from said distribution or payment by the Managers.
- 10.2. <u>Notices</u>. Except as expressly set forth to the contrary in this Agreement, all notices, requests, or consents required or permitted to be given under this Agreement must be in writing and shall be deemed to have been properly given if sent by registered or certified mail,

postage prepaid, by commercial overnight courier, by facsimile or if delivered in hand to Members at their addresses on Schedule A, or such other address as a Member may specify by notice to the Managers and to the Company or the Managers at the address of the principal office of the Company specified in Section 1.3. Whenever any notice is required to be given by law, the Certificate or this Agreement, a written waiver thereof, signed by the person entitled to notice, whether before or after the time stated therein, shall be deemed equivalent to the giving of such notice.

- 10.3. Entire Agreement; Binding Effect. This Agreement constitutes the entire agreement of the Members and the Managers relating to the Company and supersedes all prior oral or written agreements or understandings with respect to the Company. This Agreement is binding on and inures to the benefit of the parties and their respective successors, permitted assigns and legal representatives.
- 10.4. Amendment or Modification. Except as specifically provided herein, this Agreement may be amended or modified from time to time only by a written instrument signed by Members holding a majority of the Membership Interests. Notwithstanding the foregoing, any provision hereof requiring the unanimous affirmative vote or written consent of the Members may be amended or modified only by a written instrument signed by all Members.
- 10.5. Governing Law; Severability. This Agreement is governed by and shall be construed in accordance with the law of the State of New Hampshire, exclusive of its conflict-of-laws principles. In the event of a conflict between the provisions of this Agreement and any provision of the Certificate or the Act, the applicable provision of this Agreement shall control, to the extent permitted by law. If any provision of this Agreement or the application thereof to any person or circumstance is held invalid or unenforceable to any extent, the remainder of this Agreement and the application of that provision shall be enforced to the fullest extent permitted by law.
- 10.6. <u>Further Assurances</u>. In connection with this Agreement and the transactions contemplated hereby, each Member shall execute and deliver any additional documents and instruments and perform any additional acts that may be necessary or appropriate to effectuate and perform the provisions of this Agreement and those transactions, as requested by the Managers.
- 10.7. Waiver of Certain Rights. Each Member irrevocably waives any right it may have to maintain any action for dissolution of the Company or for partition of the property of the Company. The failure of any Member to insist upon strict performance of a covenant hereunder or of any obligation hereunder, irrespective of the length of time for which such failure continues, shall not be a waiver of such Member's right to demand strict compliance herewith in the future. No consent or waiver, express or implied, to or of any breach or default in the performance of any obligation hereunder shall constitute a consent or waiver to or of any other breach or default in the performance of the same or any other obligation hereunder.
- 10.8. <u>Third-Party Beneficiaries</u>. The provisions of this Agreement are not intended to be for the benefit of any creditor or other person to whom any debts or obligations are owed by,

or who may have any claim against, the Company or any of its Members or Managers, except for Members or Managers in their capacities as such. Notwithstanding any contrary provision of this Agreement, no such creditor or person shall obtain any rights under this Agreement or shall, by reason of this Agreement, be permitted to make any claim against the Company or any Member or Manager.

- 10.9. <u>Interpretation</u>. For the purposes of this Agreement, terms not defined in this Agreement shall be defined as provided in the Act; and all nouns, pronouns and verbs used in this Agreement shall be construed as masculine, feminine, neuter, singular, or plural, whichever shall be applicable. Titles or captions of Articles and Sections contained in this Agreement are inserted as a matter of convenience and for reference, and in no way define, limit, extend or describe the scope of this Agreement or the intent of any provision hereof.
- 10.10. <u>Counterparts</u>. This Agreement may be executed in any number of counterparts with the same effect as if all parties had signed the same document, and all counterparts shall be construed together and shall constitute the same instrument.

IN WITNESS WHEREOF, the parties hereto have executed this Operating Agreement under seal as of the date set forth above.

MANAGERS:

Gregory B. Carbone

MEMBERS:

Gregory B. Carbone

Emily A Carbona

BRETTON HEALTHCARE SOLUTIONS LLC

Schedule A

MANAGER

Name and Address of Manager

Gregory B. Carbone 32 Foxcross Circle Concord, NH 03301

MEMBERS

Name and Address of Members	Contribution	Membership Interest
Gregory B. Carbone 32 Foxcross Circle Concord, NH 03301	\$\$99	99%
Emily A. Carbone 32 Foxcross Circle Concord, NH 03301	\$\$1	1%

ACORD

BRETHEA-01

BMERRITT

DATE (MM/DD/YYYY)

CERTIFICATE OF LIABILITY INSURANCE

1/21/2022

THIS CERTIFICATE IS ISSUED AS A MATTER OF INFORMATION ONLY AND CONFERS NO RIGHTS UPON THE CERTIFICATE HOLDER. THIS CERTIFICATE DOES NOT AFFIRMATIVELY OR NEGATIVELY AMEND, EXTEND OR ALTER THE COVERAGE AFFORDED BY THE POLICIES BELOW. THIS CERTIFICATE OF INSURANCE DOES NOT CONSTITUTE A CONTRACT BETWEEN THE ISSUING INSURER(S), AUTHORIZED REPRESENTATIVE OR PRODUCER, AND THE CERTIFICATE HOLDER.

IMPORTANT: If the certificate holder is an ADDITIONAL INSURED, the policy(ies) must have ADDITIONAL INSURED provisions or be endorsed. If SUBROGATION IS WAIVED, subject to the terms and conditions of the policy, certain policies may require an endorsement. A statement on this certificate does not confer rights to the certificate holder in liquid for such and or remember.

tr	iis c	ertifica	ite does not	confer rights t	o the	cert	ificate holder in lieu of si		:(s).			
	DUCE							CONTACT NAME:				2
Knapton Reade & Woods Agency, Inc. 22 School Street Hillsboro, NH 03244					PHONE (A/C, No, Ext): (603	3) 464-3422	FAX (A/C, No):	(603)	464-4066			
								E-MAIL ADDRESS:		NACO CONTRACTOR OF THE PROPERTY OF THE PROPERT		
									INSURER(S) AFFO	ORDING COVERAGE		NAIC#
								INSURER A : Trave	elers Insuran	ice		19046
INSL	INSURED							INSURER B : Zurio				
Bretton Healthcare Solutions LLC 32 Foxcross Circle								INSURER C :				
								INSURER D :				
		С	oncord, NH	03301				INSURER E :				
								INSURER F :				
co	VED	AGES	1	CEE	TIEI	ATI	E NUMBER:	INSURER F.		REVISION NUMBER:		
							SURANCE LISTED BELOW	HAVE BEEN ISSUE	D TO THE INCL		THE DO	LICY BERIOD
C	IDICA ERTI	ATED. FICATE	NOTWITHST. MAY BE IS:	ANDING ANY F SUED OR MAY	REQUI PER	REM TAIN,	ENT, TERM OR CONDITIO , THE INSURANCE AFFOR LIMITS SHOWN MAY HAVE	N OF ANY CONT DED BY THE POI	RACT OR OTHE LICIES DESCRI	R DOCUMENT WITH RESP BED HEREIN IS SUBJECT	ECT TO	WHICH THIS
INSR LTR			TYPE OF INSUR	ANCE	ADDL INSD	SUBR	POLICY NUMBER	POLICY EF (MM/DD/YY)	F POLICY EXP (MM/DD/YYYY)	LIMI	rs	
A	Х		ERCIAL GENERA	ne Albertane Statemen	III	****		(WWW/DD/YY)	I) (WWW/DD/TTYY)	EACH OCCURRENCE	s	1,000,000
		CI	LAIMS-MADE	OCCUR			BIP3S330705	6/10/202	1 6/10/2022	DAMAGE TO RENTED PREMISES (Ea occurrence)	\$	300,000
										MED EXP (Any one person)	\$	5,000
										PERSONAL & ADV INJURY	\$	Excluded
	GEN	I'L AGGF	REGATE LIMIT AF	PPLIES PER:						GENERAL AGGREGATE	\$	2,000,000
		POLICY	PRO- JECT	LOC						PRODUCTS - COMP/OP AGG	\$	2,000,000
		OTHER	t:								\$	
Α	AUT	OMOBIL	E LIABILITY					- I		COMBINED SINGLE LIMIT (Ea accident)	\$	1,000,000
		ANY AL					BIP3S330705	6/10/202	1 6/10/2022			
		OWNED	ONLY	SCHEDULED AUTOS						BODILY INJURY (Per accident)	\$	
		HIRED	ONLY X	NON-OWNED AUTOS ONLY						PROPERTY DAMAGE (Per accident)	s	
		7,0100	0.1.2.1	AOTOG ONET						(* 0, 200,001,0)	s	
		UMBRE	LLA LIAB	OCCUR						EACH OCCURRENCE	s	
		EXCES	S LIAB	CLAIMS-MADE						AGGREGATE	s	
		DED	RETENTIO	NS.		122				NOONLONIE	s	
В	WOR	KERS C	OMPENSATION							PER OTH-	1	
	WORKERS COMPENSATION AND EMPLOYERS' LIABILITY ANY PROPRIETOR/PARTNER/EXECUTIVE OFFICER/MEMBER EXCLUDED? (Mandatory in NH) If yes, describe under DESCRIPTION OF OPERATIONS below				6ZZUB5R78457-6-21			3/15/2021	1 3/15/2022	E.L. EACH ACCIDENT	s	1,000,000
					N/A					E.L. DISEASE - EA EMPLOYER		1,000,000
											1000	1,000,000
	DES	CRIPTIO	NOF OPERATIO	NS below						E.L. DISEASE - POLICY LIMIT	2	
DES	PIDT	ION OF (OPERATIONS / I	OCATIONS / VEHIC	IES /A	COPI	D 101, Additional Remarks Schedu	ula may be attached if	more on see in resu	irod)		
Wor	kers	Comp	Info: Exclud	ed Officer: Gre	gory (Carb	one, Member. States Inclu	ided: NH	more space is requ	irea)		
200	-6	V-12 (10 10 10 10 10 10 10 10 10 10 10 10 10 1				-						
CE	RTIF	ICATE	HOLDER					CANCELLATIO	N			
								SHOULD AND	ETUE ADOVE	SECONIDED DOLLOISO DE O	ANIOT	LED BEFORE
		_								DESCRIBED POLICIES BE C HEREOF, NOTICE WILL		
			tate Of NH, D 29 Pleasant S	ept of Health	& Hui	man	Serv.	ACCORDANCE	WITH THE POLI	CY PROVISIONS.		11
			oncord, NH (
								AUTHORIZED REPRE	SENTATIVE			
								600		• —		
								Musar	an m	enil		

ACORD

BRETHEA-01

BMERRITT

3/14/2022

CERTIFICATE OF LIABILITY INSURANCE

THIS CERTIFICATE IS ISSUED AS A MATTER OF INFORMATION ONLY AND CONFERS NO RIGHTS UPON THE CERTIFICATE HOLDER. THIS CERTIFICATE DOES NOT AFFIRMATIVELY OR NEGATIVELY AMEND, EXTEND OR ALTER THE COVERAGE AFFORDED BY THE POLICIES BELOW. THIS CERTIFICATE OF INSURANCE DOES NOT CONSTITUTE A CONTRACT BETWEEN THE ISSUING INSURER(S), AUTHORIZED REPRESENTATIVE OR PRODUCER, AND THE CERTIFICATE HOLDER.

IMPORTANT: If the certificate holder is an ADDITIONAL INSURED, the policy(ies) must have ADDITIONAL INSURED provisions or be endorsed.

If SUBROGATION IS WAIVED, subject to the terms and conditions of the policy, certain policies may require an endorsement. A statement on this contificate does not confer rights to the certificate holder in lieu of such endorsement(s).

	certificate does not confer rights to	, tile	JUIL	moute norder in ned of st	CONTACT NAME:	•			
PRODU Knapt	on Reade & Woods Agency, Inc.					164-3422	FAX	(603)	464-4066
22 Sc	hool Street				PHONE (A/C, No, Ext): (603) 464-3422 F-MAIL ADDRESS: FAX (A/C, No): (603) 464-4066				
Hillsb	oro, NH 03244								NAIC #
							RDING COVERAGE		19046
					INSURER A : Travele	19040			
INSUR	ED .				INSURER B : Zurich				
	Bretton Healthcare Solutions	s LLC	;		INSURER C:				
	32 Foxcross Circle Concord, NH 03301				INSURER D:				
	Concord, NT 03301				INSURER E :				
				v storovenski se	INSURER F :				
				NUMBER:			REVISION NUMBER:	T	LIOV DEDICE
IND	S IS TO CERTIFY THAT THE POLICIE ICATED. NOTWITHSTANDING ANY R RTIFICATE MAY BE ISSUED OR MAY CLUSIONS AND CONDITIONS OF SUCH	EQUI PER	REMI FAIN.	ENT, TERM OR CONDITIO THE INSURANCE AFFOR	N OF ANY CONTRA DED BY THE POLIC	CT OR OTHER IES DESCRIB	R DOCUMENT WITH RESP ED HEREIN IS SUBJECT	ECTIO	WHICH THIS
INSR LTR		ADDL INSD			POLICY EFF (MM/DD/YYYY)	POLICY EXP (MM/DD/YYYY)	LIMI	TS	
	X COMMERCIAL GENERAL LIABILITY	INSU	****		(IIIII)	Lamine Street	EACH OCCURRENCE	\$	1,000,000
A1017771	CLAIMS-MADE X OCCUR			BIP3S330705	6/10/2021	6/10/2022	DAMAGE TO RENTED PREMISES (Ea occurrence)	s	300,000
				DII 00000100	3		MED EXP (Any one person)	\$	5,000
-							PERSONAL & ADV INJURY	s	Excluded
	DENII ACCRECATE LIMIT ASSISSES						GENERAL AGGREGATE	\$	2,000,000
-	POLICY PRO- LOC						PRODUCTS - COMP/OP AGG	s	2,000,000
-							PRODUCTS - COMPTOP AGG	S	
Α	OTHER:						COMBINED SINGLE LIMIT	0	1,000,000
7	ANY AUTO			BIP3S330705	6/10/2021	6/10/2022	(Ea accident) BODILY INJURY (Per person)	\$	NF2 W
-	ANY AUTO OWNED SCHEDULED AUTOS ONLY AUTOS			DIF 33330703	0/10/2021	0) 10/2022	BODILY INJURY (Per person) BODILY INJURY (Per accident		
-							PROPERTY DAMAGE (Per accident)) \$ \$	
-	HIRED AUTOS ONLY X NON-OWNED AUTOS ONLY						(Per accident)	s	
-							5.00.000.000.000		
-	UMBRELLA LIAB OCCUR						EACH OCCURRENCE	\$	
-	EXCESS LIAB CLAIMS-MADE						AGGREGATE	\$	
D.	DED RETENTION \$	-					PER OTH-	\$	
	VORKERS COMPENSATION AND EMPLOYERS' LIABILITY Y / N	N/A		6ZZUB5R78457-6-22	3/15/2022	3/15/2023		1.	1,000,000
A	NY PROPRIETOR/PARTNER/EXECUTIVE Y DEFICER/MEMBER EXCLUDED? Mandatory in NH)				0/10/2022	5///5//	E.L. EACH ACCIDENT	\$	1,000,000
	Mandatory in NH) f yes, describe under						E.L. DISEASE - EA EMPLOYE		1,000,000
i	DESCRIPTION OF OPERATIONS below		_				E.L. DISEASE - POLICY LIMIT	\$.,,,,,,,,
		111							
DESCR Works	RIPTION OF OPERATIONS / LOCATIONS / VEHIC ers Comp Info: Excluded Officer: Gre	LES (A	ACORI Carb	D 101, Additional Remarks Sched one, Member. States Inclu	ule, may be attached if mo uded: NH	re space is requi	red)		
CER	TIFICATE HOLDER				CANCELLATION				
State Of NH, Dept of Health & Human Serv. 129 Pleasant Street					SHOULD ANY OF THE ABOVE DESCRIBED POLICIES BE CANCELLED BEFORE THE EXPIRATION DATE THEREOF, NOTICE WILL BE DELIVERED IN ACCORDANCE WITH THE POLICY PROVISIONS.				
	Concord, NH 03301-3857				AUTHORIZED REPRESENTATIVE				
					Elizabi	oh m	and II		